

# MIT Open Access Articles

Tasks, Automation, and the Rise in U.S. Wage Inequality

The MIT Faculty has made this article openly available. *Please share* how this access benefits you. Your story matters.

**Citation:** Acemoglu, Daron and Restrepo, Pascual. 2022. "Tasks, Automation, and the Rise in U.S. Wage Inequality." 90 (5).

As Published: 10.3982/ecta19815

Publisher: The Econometric Society

Persistent URL: https://hdl.handle.net/1721.1/146052

**Version:** Final published version: final published article, as it appeared in a journal, conference proceedings, or other formally published context

Terms of use: Creative Commons Attribution-NonCommercial-NoDerivs License



# TASKS, AUTOMATION, AND THE RISE IN U.S. WAGE INEQUALITY

DARON ACEMOGLU Department of Economics, MIT

#### PASCUAL RESTREPO Department of Economics, Boston University

We document that between 50% and 70% of changes in the U.S. wage structure over the last four decades are accounted for by relative wage declines of worker groups specialized in routine tasks in industries experiencing rapid automation. We develop a conceptual framework where tasks across industries are allocated to different types of labor and capital. Automation technologies expand the set of tasks performed by capital, displacing certain worker groups from jobs for which they have comparative advantage. This framework yields a simple equation linking wage changes of a demographic group to the task displacement it experiences. We report robust evidence in favor of this relationship and show that regression models incorporating task displacement explain much of the changes in education wage differentials between 1980 and 2016. The negative relationship between wage changes and task displacement is unaffected when we control for changes in market power, deunionization, and other forms of capital deepening and technology unrelated to automation. We also propose a methodology for evaluating the full general equilibrium effects of automation, which incorporate induced changes in industry composition and ripple effects due to task reallocation across different groups. Our quantitative evaluation explains how major changes in wage inequality can go hand-in-hand with modest productivity gains.

KEYWORDS: Tasks, automation, productivity, technology, inequality, wages.

# 1. INTRODUCTION

WAGE INEQUALITY has risen sharply in the United States and other industrialized economies over the last four decades.<sup>1</sup> Figure 1 depicts some of the most salient changes in the U.S. wage structure since 1980: while the real wages of workers with a post-graduate degree rose, the real wages of low-education workers fell or remained stagnant. The real earnings of men without a high-school degree are now 15% lower than they were in 1980.

This paper proposes a new approach for thinking about wage inequality. In our theory, shifts against less skilled workers result from technologies that *automate* and thus displace workers from tasks they used to perform. Our main contribution is to develop a general version of this theory and show how it can be applied to quantify the effects of automation on wages and inequality. Based on this approach, we document that between 50% and 70% of the overall changes in the U.S. wage structure over the last four decades are driven by automation. In particular, automation reduced the relative, and in some cases real, wages of workers specializing in routine tasks in industries undergoing rapid

<sup>1</sup>See Goldin and Katz (2008), Acemoglu and Autor (2011), and Autor (2019) for overviews.

© 2022 The Authors. Econometrica published by John Wiley & Sons Ltd on behalf of The Econometric Society. Daron Acemoglu is the corresponding author on this paper. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Daron Acemoglu: daron@mit.edu

Pascual Restrepo: pascual@bu.edu

We thank David Autor, Gino Gancia, Maarten Goos, Patrick Kline, Thomas Lemieux, Jonas Loebbing, Richard Rogerson, Esteban Rossi-Hansberg, Stephen Ross, Jesse Shapiro, and three anonymous referees for their comments and suggestions, and seminar participants at various universities and conferences for very helpful comments. We thank Eric Donald and Nils Lehr for excellent research assistance. We also gratefully acknowledge financial support from Google, the Hewlett Foundation, Microsoft, the NSF, Schmidt Sciences, the Sloan Foundation, and the Smith Richardson Foundation. After the online edition was first posted, two sentences inadvertently left out from the paper have been added.



FIGURE 1.—Cumulative growth of real hourly wages by gender and education (from Autor (2019)).

automation (such as those working in blue-collar jobs in manufacturing industries that introduced numerical-controlled machinery and industrial robots, or those in clerical tasks in industries that introduced software-based automation). In contrast, worker groups that were not displaced from their tasks, such as those with a post-graduate degree or women with a college degree, enjoyed wage gains.

Our framework models the allocation of a range of tasks across industries to capital and demographic groups, each with a different comparative advantage. Technological progress can increase the productivity of some demographic groups (e.g., skill-biased technological change, SBTC, can augment the productivity of groups with higher education); it can raise the productivity of capital in its current tasks; and most importantly, it can automate work—which means that the productivity of machines and algorithms increases in tasks previously allocated to workers, thus expanding the range of tasks performed by capital. Our model clarifies the distinct effects of these technological changes: automation displaces workers from tasks where they had comparative advantage, reducing their relative wages and even possibly their real wages.<sup>2</sup> In contrast, technologies directly improving the productivity of skilled labor do not involve any displacement and always increase the wages of unskilled workers. In addition, their effects on inequality and factor shares are ambiguous and depend on elasticities of substitution.

The most important contribution of our framework is to provide a methodology for empirically investigating these predictions. At the center of this contribution is a simple equation that relates wage changes of a worker group to the direct *task displacement* it experiences—a measure summarizing the share of tasks this group of workers loses directly to automation. We show that a group's direct task displacement can be measured as a (weighted) average of automation-driven labor share declines across industries where it specializes in tasks that can be automated.

<sup>&</sup>lt;sup>2</sup>We define automation technologies as any technology that enables machines or algorithms to perform tasks previously allocated to humans (which thus leads to the displacement of workers from these tasks). Note, however, that task displacement does not need to be associated with "job loss," and can take the form of a worker being reallocated within the same firm or a decline in hiring of new workers.

The second part of the paper documents a robust negative reduced-form relationship between direct task displacement and real wages across groups of workers. For this empirical exercise, we focus on 500 demographic groups defined by education, gender, age, race, and native/immigrant status. We identify tasks that can be automated with those that are routine (as classified in Acemoglu and Autor (2011)). Our first measure of task displacement exploits observed industry labor share declines, which in our framework are closely connected to automation.<sup>3</sup> Our preferred measure of task displacement uses information on automation-driven industry labor share declines, which we estimate using data on the adoption of robots, specialized software, and dedicated machinery across industries. These proxies of automation account for 45% of the observed changes in industry labor shares from 1987 to 2016. Using both measures, we find a strong association between task displacement and wages. In our baseline regressions, direct task displacement explains 50-70% of the changes in wage structure across groups between 1980 and 2016. This holds regardless of whether we control for standard forms of SBTC (e.g., allowing the productivity of workers to evolve as a function of their education levels and gender), which account for 10% of the changes in the wage structure. Consistent with the notion that task displacement reflects changes in labor demand, we also estimate negative effects of task displacement on employment.

The relationship between task displacement and wages is unaffected when we control for other potential determinants of industry labor shares and earnings, such as changes in industry concentration and markups, Chinese import competition, and deunionization, and these factors themselves do not appear to have a sizable effect on the U.S. wage structure. Our results also remain unchanged when we control for other forms of capital deepening and TFP growth unrelated to automation. This shows that our estimates are driven by automation and not by other forms of technological progress or capital deepening. Finally, we also show that offshoring-induced task displacement has similar effects on wages, though offshoring accounts for a smaller share of the observed changes in task displacement and the wage structure than automation.<sup>4</sup>

Although our reduced-form analysis documents a strong negative relationship between direct task displacement and *relative* wage changes across worker groups, it misses three indirect channels via which automation affects wages in general equilibrium. First, in our regressions, the common effect of productivity increases on wages goes into the intercept, and so our results are not informative about real wage *level* changes. Second, because automation concentrates in some industries, it will change the industry composition of the economy, which in turn shifts the demand for different types of workers. Third, our reduced-form evidence focuses on the direct task displacement experienced by each group of workers, but does not account for *ripple effects*, which result from displaced workers competing against others for non-automated tasks, bidding down wages and spreading automation's effects more broadly in the population.

1975

<sup>&</sup>lt;sup>3</sup>There are many determinants of industry labor shares, and we explore and control for their effects later. See Elsby, Bart, and Ayşegül (2013), Karabarbounis and Neiman (2013), Piketty (2014), Dao, Das, and Koczan (2019), and Hubmer (2020) on the decline of the labor share; Acemoglu, Lelarge, and Restrepo (2020) and Acemoglu and Restrepo (2020) on the role of automation in labor share declines; De Loecker, Eeckhout Jan, and Unger (2020) on the role of rising markups; and Autor, Dorn, Katz, Patterson, and Van Reenen (2020) and Hubmer and Restrepo (2021) on superstar firms.

<sup>&</sup>lt;sup>4</sup>In additional empirical exercises, we find similar results when we exploit regional variation in specialization patterns (instead of national variation in specialization patterns across groups) to compute our task displacement measures, or when we look at different subperiods.

The third part of the paper undertakes a quantitative exploration of these general equilibrium mechanisms and estimates the full implications of automation for the wage structure, real wages, TFP, output, and the industry composition of the economy. Our framework provides explicit formulas to compute these general equilibrium effects as functions of direct task displacement as well as the cost savings from automation, industry demand elasticities, and a *propagation matrix* representing the strength of ripple effects between different groups of workers (i.e., how much the displacement of group g affects the wage of group g'). The propagation matrix can be estimated by parameterizing group-level interactions as functions of the distance between groups. We combine these estimates with a standard parameterization of demand across industries, available estimates of cost savings from automation, and our measures of direct task displacement to compute the full general equilibrium implications of automation.

We find that automation—incorporating general equilibrium effects—accounts for 50% of the changes in the wage structure during this period and explains 80% of the rise in the college premium. At the same time, we estimate that automation reduced the real wage of high-school dropout men by 8.8% and high-school dropout women by 2.3%. These sizable distributional effects are accompanied by small increases in the average wage level, GDP, and TFP. For example, we find that automation accounts only for a (cumulative) 3.4% increase in TFP between 1980 and 2016. We conclude that stagnant and declining real wages and slow productivity growth can go hand-in-hand in the presence of rapid automation.

Our work contributes to various literatures. First, our conceptual framework builds on previous task models, in particular, Zeira (1998), Acemoglu and Zilibotti (2001), Autor, Levy, and Murnane (2003), Acemoglu and Autor (2011), and Acemoglu and Restrepo (2018), as well as Grossman and Rossi-Hansberg's (2008) model of offshoring. Relative to these papers, our main innovation is our methodology for measuring and estimating the effects of task displacement on wages. The central element of this methodology is the explicit formulas linking wage changes to task displacement, which underpin our empirical work. As part of this contribution, we also develop a general version of existing models of automation and offshoring, in which there are many sectors, many tasks within each sector, and a large number of demographic groups with flexible comparative advantage across tasks and sectors (see also Jackson and Kanik (2020) for a complementary framework to study the implications of automation on inequality).

Most closely related to our paper is Autor, Levy, and Murnane (2003), who also focus on the effects of technologies automating routine tasks and complementing non-routine workers. Our paper can be seen as a generalization of their conceptual framework and derives an explicit relationship between task displacement and wages, which we use in both reduced-form and structural estimation. Other empirical explorations of the consequences of automation include Acemoglu and Autor (2011), Graetz and Michaels (2018), and Acemoglu and Restrepo (2020). These works do not estimate the direct and/or general equilibrium effects of task displacement on the wage structure. Acemoglu and Restrepo (2020), for example, estimated the causal impacts of industrial robots on local employment and wages, but did not look at their effects on the national wage structure, which is our main focus here. It is also important to recall that industrial robots are one of several automation technologies adopted in the U.S. economy over the last four decades.<sup>5</sup>

<sup>&</sup>lt;sup>5</sup>Our findings also complement works on job polarization, such as Goos and Manning (2007), Goos, Manning, and Salomons (2014), and Autor and Dorn (2013). We document that groups most affected by task displacement are in the middle of the wage distribution, thus linking task displacement to polarization. Other

Second, our work builds on but fundamentally departs from the traditional literature on SBTC. This literature starts with an aggregate production function of the form  $F(A_HH, A_LL)$ , where H and L are high-skill and low-skill labor, and  $A_H$  and  $A_L$  represent technologies augmenting these workers. SBTC corresponds to technology becoming more favorable to high-skill workers (e.g., a bigger increase in  $A_H$  than in  $A_L$ , provided that F has an elasticity of substitution greater than 1). Several works, including Bound and Johnson (1992), Katz and Murphy (1992), Krueger (1993), Autor, Katz, and Krueger (1998), and Card and Lemieux (2001), have explored the evolution of between-group wage inequality in response to changes in factor supplies and skill-augmenting technologies (increases in  $A_H$ ). We differ from this literature in a number of ways. Most importantly, our focus is on automation technologies displacing certain groups of workers and *not* on technologies complementing high-skill workers. Our theoretical framework and our empirical results point to a limited role for factor-augmenting technologies in shaping the U.S. wage structure over the last four decades.<sup>6</sup>

Third, our work builds on and complements the literature exploring the effects of lower equipment and computer prices on wage inequality through capital-skill complementarity. This literature posits an aggregate production function of the form F(K, H, L), in which capital (or equipment) K directly complements skilled workers. These ideas go back to Griliches (1969), and their implications for U.S. wage inequality have been explored in Krusell, Ohanian, Ríos-Rull, and Violante (2000) and Burstein, Morales, and Vogel (2019). As with the SBTC literature, the main mechanism via which technology and capital impact inequality in this literature is through complementarity—thus without any role for task displacement. We clarify the distinction between automation and the capital-skill complementarity studied in this literature, and show that automation has a powerful impact on inequality even when there are no direct capital-skill complementarities.

The rest of the paper is organized as follows. The next section introduces our framework and derives the key equations for our empirical work. Section 3 presents our data and measurement strategy. Section 4 presents the reduced-form evidence. Section 5 explores the general equilibrium effects of automation. Section 6 concludes. Appendix A of the Supplemental Material (Acemoglu and Restrepo (2022)) contains proofs and reports our main robustness checks. Appendix B, which is available upon request, provides additional theoretical results and robustness checks for our quantitative exercise.

### 2. CONCEPTUAL FRAMEWORK: TASKS, WAGES, AND INEQUALITY

We start with a single-sector model that illustrates how automation and other technologies affect wages. We then move to our multi-sector model and formally derive the task displacement measure we use in our empirical work.

1977

papers studying the decline of routine occupations and broader changes in occupational structure include, Johnson and Keane (2013), Lee, Tim, and Shin (2017), Gregory, Salomons, and Zierahn (2018), Bárány and Siegel (2020), Jaimovich, Saporta-Eksten, Siu, and Yedid-Levi (2020), Atalay, Phongthiengtham, Sotelo, and Tannenbaum (2020), and Caunedo, Jaume, and Keller (2021).

<sup>&</sup>lt;sup>6</sup>In principle, one could develop a more general form of SBTC whereby technological change increases  $A_H$  and *simultaneously* reduces  $A_L$ . This more general version would capture some displacement effects, though without microfoundations it is not clear why technological *progress* would make some workers *less* productive. Our theory can be viewed as providing microfoundations for this type of general SBTC model.

#### 2.1. Single Sector

*Environment and Equilibrium.* Output is produced by combining a mass M of tasks in a set  $\mathcal{T}$  using a constant elasticity of substitution (CES) aggregator with elasticity  $\lambda \ge 0$ ,

$$y = \left(\frac{1}{M}\int_{\mathcal{T}} \left(M \cdot y(x)\right)^{\frac{\lambda-1}{\lambda}} \cdot dx\right)^{\frac{\Lambda}{\lambda-1}},$$

where *x* indexes tasks. For example, producing a shirt requires the completion of a range of tasks, including designing it; cleaning, carding, combing, and spinning the fibers; weaving, knitting, and bonding of yarn; dying, chemical processing, and finishing; marketing and advertising; transport; and various wholesale and retail tasks.

The key economic decision in this model is how to perform these tasks. Each task can be produced using capital or different types of labor indexed by g (where  $g \in \mathcal{G} = \{1, 2, ..., G\}$ ):

$$y(x) = A_k \cdot \psi_k(x) \cdot k(x) + \sum_{g \in \mathcal{G}} A_g \cdot \psi_g(x) \cdot \ell_g(x).$$

Here,  $\ell_g(x)$  is the amount of labor of type g allocated to task x, while k(x) is the amount of task-specific capital produced for and assigned to this task. The  $A_k$  and  $A_g$  terms represent standard factor-augmenting technologies, which make factors uniformly more productive in all tasks. More importantly, productivity has a task-specific component, represented by the functions  $\psi_k(x)$  and  $\{\psi_g(x)\}_{g\in\mathcal{G}}$ , which determine comparative advantage and specialization patterns. Task-specific productivity is zero for factors that cannot perform a task.

Capital for performing task x, k(x), is produced using the final good at a constant marginal cost 1/q(x). Net output, which is equal to consumption, is therefore obtained by subtracting the production cost of capital goods from output:

$$c = y - \int_{\mathcal{T}} \left( k(x)/q(x) \right) \cdot dx.$$

Labor is supplied inelastically, and we denote the total supply of labor of type g by  $\ell_g$ .

A market equilibrium is defined as an allocation of tasks to factors and a production plan for capital goods that maximizes consumption. Given a supply of labor  $\ell = (\ell_1, \ell_2, \dots, \ell_G)$ , a market equilibrium is specified by wages  $\boldsymbol{w} = (w_1, w_2, \dots, w_G)$ , capital production decisions k(x), and an allocation of labor to tasks,  $\ell_g(x)$  such that: (i) the allocation of tasks to factors minimizes costs; (ii) capital production decisions maximize net output; and (iii) the markets for capital goods and different types of labor clear. We set the final good as the numeraire, so that the  $w_g$ 's correspond to real wages. Throughout, when a task can be produced at the exact same unit cost by different factors, we assume it is allocated to capital or to the type of labor with the higher index, and we also assume that each factor has a strict comparative advantage for some tasks.<sup>7</sup>

<sup>&</sup>lt;sup>7</sup>The tie-breaking rule simplifies our exposition and has no substantive effect on equilibrium, except that in Proposition 1, it enables us to state that the equilibrium is unique (rather than "essentially" unique at these non-generic points of cost equality). The second part of the assumption is to ensure that an equilibrium satisfying this tie-breaking rule always exists (an equilibrium without this rule always exists). Formally, this assumption requires that for any positive measure subset of tasks  $\mathcal{T}' \subset \mathcal{T}$  and for any g and g' (with the convention that g = 0 stands for k),  $\psi_{g'}(x)/\psi_g(x)$  is not constant for all  $x \in \mathcal{T}'$ .

*Task Shares.* Cost minimization and our tie-breaking rule imply that each task is produced by a single factor. Let  $\mathcal{T}_g$  represent the set of tasks allocated to labor of type g, and  $\mathcal{T}_k$  the set of tasks allocated to capital. These sets are equilibrium objects that satisfy

$$\mathcal{T}_{g} = \left\{ x : \frac{w_{g}}{\psi_{g}(x) \cdot A_{g}} \le \frac{w_{j}}{\psi_{j}(x) \cdot A_{j}} \text{ for } j < g; \\ \frac{w_{g}}{\psi_{g}(x) \cdot A_{g}} < \frac{w_{j}}{\psi_{j}(x) \cdot A_{j}}, \frac{1}{\psi_{k}(x) \cdot q(x) \cdot A_{k}} \text{ for } j > g \right\},$$
$$\mathcal{T}_{k} = \left\{ x : \frac{1}{\psi_{k}(x) \cdot q(x) \cdot A_{k}} \le \frac{w_{j}}{\psi_{j}(x) \cdot A_{j}} \text{ for all } j \right\}.$$

Given an allocation of tasks to factors, we define

$$\Gamma_g(\boldsymbol{w}, \Psi) = \frac{1}{M} \int_{\mathcal{T}_g} \psi_g(x)^{\lambda-1} \cdot dx \quad \text{and} \quad \Gamma_k(\boldsymbol{w}, \Psi) = \frac{1}{M} \int_{\mathcal{T}_k} (\psi_k(x) \cdot q(x))^{\lambda-1} \cdot dx.$$

The quantities  $\Gamma_g$  and  $\Gamma_k$ , which we refer to as the *task shares* of workers of type g and capital, respectively, give the measure of the set of tasks allocated to a factor weighted by the "importance" of the tasks.<sup>8</sup> Task shares depend on the sets  $\mathcal{T}_g$  and  $\mathcal{T}_k$ , and thus on wages, factor-augmenting technologies, and task productivities. Consequently, they are functions of the vectors of wages  $\boldsymbol{w}$  and technology  $\Psi = (\{\psi_k(x), \psi_g(x), q(x)\}_{x \in \mathcal{T}}, A_k, \{A_g\}_{g \in \mathcal{G}})$ , but we omit this dependence when it causes no confusion.

The next proposition characterizes the equilibrium, and expresses factor prices, shares, and output as functions of task shares. Because production in this economy is "round-about" (capital is produced linearly from the final good), output can be infinite. In Appendix A-2 of the Supplemental Material, we derive an Inada condition that ensures finite output (in the one-sector case, this condition implies  $A_k^{\lambda-1} \cdot \Gamma_k < 1$ ), and we assume throughout that it is satisfied.

**PROPOSITION 1**—Equilibrium: There is a unique equilibrium. In this equilibrium, output, wages, and the capital share in GDP,  $s^{K}$ , can be expressed as functions of task shares:

$$y = \left(1 - A_k^{\lambda - 1} \cdot \Gamma_k\right)^{\frac{\lambda}{1 - \lambda}} \cdot \left(\sum_{g \in \mathcal{G}} \Gamma_g^{\frac{1}{\lambda}} \cdot (A_g \cdot \ell_g)^{\frac{\lambda - 1}{\lambda}}\right)^{\frac{\lambda}{\lambda - 1}},\tag{1}$$

$$w_{g} = \left(\frac{y}{\ell_{g}}\right)^{\frac{1}{\lambda}} \cdot A_{g}^{\frac{\lambda-1}{\lambda}} \cdot \Gamma_{g}^{\frac{1}{\lambda}} \quad \text{for all } g \in \mathcal{G},$$

$$\tag{2}$$

$$s^{K} = A_{k}^{\lambda - 1} \cdot \Gamma_{k}. \tag{3}$$

The proposition establishes that task shares—the  $\Gamma_g$ 's and  $\Gamma_k$ —are the key objects summarizing the distributional effects of technology. Equation (1) shows that output can be represented as a CES aggregate of different types of labor and capital, with elasticity of substitution  $\lambda$ . However, this representation differs from the standard CES production

<sup>&</sup>lt;sup>8</sup>In particular, this importance weight depends on the revenue share of the task in total costs, and hence the productivity of the factor performing the task has an exponent equal to the elasticity of substitution minus 1.

function for three reasons. First, the distribution parameters, which are exogenous in the standard CES, are now endogenous and are given by the task shares, the  $\Gamma_g$ 's. They are functions of not just factor prices (via the dependence of the sets  $\mathcal{T}_g$  and  $\mathcal{T}_k$  on factor prices), but also technology. Second, despite appearances, the elasticity of substitution between factors is *not* equal to  $\lambda$ , but  $\sigma \ge \lambda$ . The exact value of  $\sigma$  depends on endogenous substitution taking place as tasks are reallocated (again captured by changes in the sets  $\mathcal{T}_g$  and  $\mathcal{T}_k$ , or variations in the  $\Gamma_g$ 's and  $\Gamma_k$  in response to factor prices). Finally, the term  $1 - A_k^{\lambda-1} \cdot \Gamma_k > 0$  accounts for the roundabout nature of production.

Equation (2) is intuitive: real wages are given by the marginal product of each type of labor, which is a function of output per worker (raised to the power  $1/\lambda$ ) and the factor-augmenting technology,  $A_g$  (raised to the power  $(\lambda - 1)/\lambda$ ). More novel and central to our empirical strategy is that wages also depend directly on task shares, the  $\Gamma_g$ 's, highlighting a key aspect of our model: the real wage of a factor is linked to its task share.

Although task shares are endogenous objects, Proposition 1 is useful because it clarifies how the impact of automation technologies on equilibrium outcomes work via their influence on task shares. In particular, automation impacts equilibrium prices and quantities by reallocating tasks away from labor and thus reducing the  $\Gamma_g$ 's. Building on this insight, we next show that the effects of automation on wages can be studied by tracing its impact on task shares.

*The Effects of Technology.* To understand the distinct effects of automation, it is useful to first contrast them with those of other technologies:

■ Factor-augmenting technologies: represented by higher  $A_g$  or  $A_k$ . Factor-augmenting technologies have been the focus of much of the macro and labor literatures. They are qualitatively different from automation technologies and arguably a significant abstraction, since there are no examples of technologies that increase factor productivity in *all* or even *most* tasks.

■ Productivity-deepening technologies: these correspond to increases in the productivity of a factor at tasks it currently performs—represented by an increase in  $\psi_g(x)$  for  $x \in \mathcal{T}_g$ in the case of labor or in  $\psi_k(x)$  for  $x \in \mathcal{T}_k$  in the case of capital. For example, we may have improvements in the tools used by workers to perform one of their tasks (think of GPS making drivers better at navigation), or upgrades in capital equipment used to produce the same task. Formally, we consider infinitesimal increases in  $\psi_g(x)$  for  $x \in \mathcal{T}_g$ , and define the direct effect of these changes on group g's task share as

$$d\ln\Gamma_g^{\text{deep}} = \frac{1}{M} \int_{\tau_g} \frac{\psi_g(x)^{\lambda-1}}{\Gamma_g} \cdot d\ln\psi_g(x) \, dx. \tag{4}$$

 $d \ln \Gamma_k^{\text{deep}}$  is defined similarly for capital.

• Automation and offshoring: automation corresponds to increases in the productivity of capital (or reductions in the cost of producing this type of capital) at tasks previously assigned to labor and leads to the displacement of workers from these tasks. Examples of automation technologies include numerical control machinery or industrial robots taking over tasks from blue-collar workers or the introduction of specialized software automating various back-office and clerical tasks. Offshoring also leads to the displacement of workers and can be incorporated into this framework by assuming that tasks can be performed abroad and imported in exchange of 1/q(x) units of the final good (see also Grossman and Rossi-Hansberg (2008)).

We model automation as a discrete increase in the productivity of capital in an infinitesimal set of tasks  $\mathcal{D}_g \subseteq \mathcal{T}_g$  (previously performed by workers of group g) such that capital now outperforms labor in these tasks. We define two objects that summarize the effects of automation. The first is the (*direct*) task displacement experienced by g:

$$d\ln\Gamma_g^{\text{auto}} = \frac{\frac{1}{M} \int_{\mathcal{D}_g} \psi_g(x)^{\lambda-1} dx}{\frac{1}{M} \int_{\mathcal{T}_g} \psi_g(x)^{\lambda-1} dx}.$$
(5)

This measure represents the direct reduction in group g's task share due to automation. We emphasize that this is automation's "direct" impact to highlight that it depends only on the underlying improvements in automation technology (increase in capital productivity in tasks previously performed by group g) and to distinguish it from its indirect impact that incorporates *ripple effects* that result from the reallocation of tasks across factors in response to changes in equilibrium prices. Second, we define *cost savings* from automating these tasks as

$$\pi_g = \frac{\frac{1}{M} \int_{\mathcal{D}_g} \psi_g(x)^{\lambda - 1} \cdot \pi_g(x) \, dx}{\frac{1}{M} \int_{\mathcal{D}_g} \psi_g(x)^{\lambda - 1} \, dx}$$

where  $\pi_g(x)$  denotes the cost reduction from automating task  $x \in \mathcal{D}_g$ .<sup>9</sup>  $\pi_g$  is also a function of the underlying technology (capital productivity in the tasks in  $\mathcal{D}_g$  after the change in technology).

Figure 2 depicts the effects of productivity deepening and automation on the allocation of tasks to factors. The *direct* effects in equations (4) and (5) are shown with the shaded areas (corresponding to the tasks where the productivity of capital or labor increased), while the induced ripple effects, which alter task shares of worker groups that are not themselves directly impacted by new technologies, are depicted with the dashed curves.

We now characterize the implications of these technologies, while abstracting from ripple effects, which allows us to illustrate their direct impacts and derive a simple estimating equation. The following assumption rules out ripple effects and is maintained until Section 5, where we characterize and estimate the full general equilibrium effects of automation on the wage structure:

- ASSUMPTION 1: 1. Workers can only produce non-overlapping sets of tasks (i.e.,  $\psi_g(x) > 0$  only if  $\psi_{g'}(x) = 0$  for all  $g' \neq g$ ).
- 2.  $\psi_k(x) > \underline{\psi}$  and  $q(x) > \overline{q}$  for all  $x \in S = \{x : \psi_k(x) > 0\}$ , where the constants  $\underline{\psi}$  and  $\overline{q}$  are defined such that, in this case,  $\mathcal{T}_k = S$ .

The first part of the assumption imposes that each task can be performed by at most one type of labor, which ensures that a group displaced from the tasks it specializes in cannot in turn displace other workers from their tasks. The second part imposes that capital

<sup>&</sup>lt;sup>9</sup>This cost saving from automating task x is in turn given as  $\pi_g(x) = \frac{1}{\lambda - 1} \left[ \left( w_g \frac{A_k \cdot q(x) \cdot \psi_k(x)}{A_g \cdot \psi_g(x)} \right)^{\lambda - 1} - 1 \right] > 0$ , where the expression is evaluated at the new level of capital productivity and initial equilibrium wages.



FIGURE 2.—The direct effects of technology and ripple effects. The left panel shows the effects of an increase of  $d \ln \Gamma_g^{\text{deep}}$  in the productivity of group g in tasks in  $\mathcal{T}_g$ . The right panel depicts the effects of automation technologies that reduce the task share of worker g by  $d \ln \Gamma_g^{\text{auto}}$ .

productivity is high enough and the cost of capital is low enough that all tasks in the set  $S = \{x : \psi_k(x) > 0\}$ , where capital has positive productivity, will be allocated to capital, that is,  $T_k = S$  (see Appendix B-1 for details and a derivation of these thresholds).

The next proposition characterizes the implications of these technologies for wages, TFP, and output in terms of their direct effects on task shares and cost savings from automation.

PROPOSITION 2—Technology Comparative Statics: Consider a change in technology (such as factor-augmenting, productivity-deepening, and automation). The impact on real wages, TFP, output, and the capital share are

$$d\ln w_g = \frac{1}{\lambda} d\ln y + \frac{\lambda - 1}{\lambda} d\ln \tilde{A}_g - \frac{1}{\lambda} d\ln \Gamma_g^{\text{auto}},\tag{6}$$

$$d\ln y = \frac{1}{1 - s^{K}} \cdot \left( d\ln \mathrm{tfp} + s^{K} \cdot d\ln s^{K} \right),\tag{7}$$

$$d\ln \mathrm{tfp} = \sum_{g \in \mathcal{G}} s_g^L \cdot d\ln \tilde{A}_g + s^K \cdot d\ln \tilde{A}_k + \sum_{g \in \mathcal{G}} s_g^L \cdot d\ln \Gamma_g^{\mathrm{auto}} \cdot \pi_g, \tag{8}$$

$$d\ln s^{\kappa} = (\lambda - 1) \cdot d\ln \tilde{A}_{\kappa} + \frac{1}{s^{\kappa}} \cdot \sum_{g \in \mathcal{G}} s_{g}^{L} \cdot d\ln \Gamma_{g}^{\text{auto}} \cdot \left(1 + (\lambda - 1) \cdot \pi_{g}\right), \quad , \qquad (9)$$

where  $d \ln \tilde{A}_g = d \ln A_g + d \ln \Gamma_g^{\text{deep}}$ ,  $d \ln \tilde{A}_k = d \ln A_k + d \ln \Gamma_k^{\text{deep}}$ , and  $s_g^L = w_g \cdot \ell_g / y$  is the share of group g in GDP.

Let us first consider factor-augmenting and productivity-deepening technologies that make workers (or capital) more productive at their current tasks. With no ripple effects, factor-augmenting and productivity-deepening technologies have identical implications, summarized by the terms  $d \ln \tilde{A}_g$  and  $d \ln \tilde{A}_k$ . Equation (6) gives their impact on the wage structure. The real wage of group g increases due to productivity gains, represented by the expansion of output,  $d \ln y$ . These technologies further affect relative wages through the term  $\frac{\lambda-1}{\lambda} \cdot d \ln \tilde{A}_g$ , whose sign depends on whether the elasticity of substitution between type g labor and other factors,  $\lambda$ , is greater than or less than 1.<sup>10</sup> This ambiguous impact is rooted in the fact that technologies that make workers from group g more productive simultaneously lower the price of the tasks these workers produce. When  $\lambda > 1$ , the first effect dominates, and technologies making a group of workers more productive will raise their relative wages. This is the standard mechanism emphasized in the SBTC literature (e.g., Katz and Murphy (1992)). Technologies increasing the productivity of group g raise the wage of all other workers (and technologies increasing the productivity of capital at its current tasks raise all wages). This is the reason theories that emphasize skill-biased technologies or capital-skill complementarities have a hard time accounting for the stagnant or decreasing wages of unskilled workers (see Acemoglu and Autor (2011)).

The impact of factor-augmenting and productivity-deepening technologies on TFP can be computed from (8) as  $\sum_{g \in \mathcal{G}} s_g^L \cdot d \ln \tilde{A}_g + s^K \cdot d \ln \tilde{A}_k$ . This formula, which follows from Hulten's theorem, has a simple envelope logic: a 1% increase in the productivity of all workers in group g leads to an increase in TFP of  $s_g^L \%$ . Likewise, a 1% increase in the productivity of capital at all tasks leads to an increase in TFP of  $s^K \%$ . Relative to their modest effects on the wage structure (especially for values of  $\lambda$  close to 1), these technologies have large productivity effects. If factor-augmenting and productivity-deepening technologies were at the root of changes in the wage structure, we should see sizable TFP gains.

These results contrast with the implications of automation, whose impact on wages in (6) is

$$\frac{1}{\lambda}d\ln y - \frac{1}{\lambda}d\ln\Gamma_g^{\text{auto}}.$$

The first term is once again the productivity effect, which raises the wages of all workers. More novel and important for our purposes is the second term, which shows that a group's real wage change depends on the task displacement it experiences; this is independent of whether  $\lambda \leq 1$ . This negative displacement effect is the defining feature of automation technologies.

The implications of automation for TFP and factor shares are distinct from those of other technologies as well. The change in TFP is now  $\sum_{g \in G} s_g^L \cdot d \ln \Gamma_g^{\text{auto}} \cdot \pi_g$ . If cost savings from automation,  $\pi_g$ , are modest, automation could have a sizable impact on the wage structure via task displacement and still bring only small aggregate productivity gains. In this case, the displacement effect can outweigh the productivity effect and the real wage for displaced groups can decline, as we will see in our general equilibrium analysis.

Equation (9) also shows that automation always increases the capital share and reduces the labor share of value added—an observation that will be at the core of our measurement approach, in Section 2.4. This, too, is in stark contrast to what one would get

<sup>&</sup>lt;sup>10</sup>In the presence of ripple effects, the impact of  $A_g$  on group g wages is  $\frac{\sigma_g - 1}{\sigma_g} d \ln A_g$ , where  $\sigma_g = \lambda \cdot \frac{1}{1 + \partial \ln \Gamma_g / \partial \ln w_g}$  is the elasticity of substitution between group g and other workers. Because an increase in  $A_g$  expands the set of tasks performed by group  $g, \sigma_g \ge \lambda$ . Under Assumption 1, however, there are no ripple effects, and thus  $\sigma_g = \lambda$ .

from factor-augmenting and productivity-deepening technologies, whose impact on factor shares depends on whether  $\lambda \leq 1$ .

# 2.2. Full Model: Multiple Sectors

Our full model generalizes the one-sector setup in the previous subsection. There are multiple industries indexed by  $i \in \mathcal{I} = \{1, 2, ..., I\}$ . Output in industry *i* is produced by combining the tasks in some set  $\mathcal{T}_i$ , with measure  $M_i$ , using a CES aggregator with elasticity  $\lambda \geq 0$ :

$$y_i = A_i \cdot \left(\frac{1}{M_i} \int_{\mathcal{T}_i} (M_i \cdot y(x))^{\frac{\lambda-1}{\lambda}} \cdot dx\right)^{\frac{\lambda}{\lambda-1}},$$

where x again indexes tasks and  $A_i$  is a Hicks-neutral productivity term.  $\mathcal{T}_{gi}$  denotes the set of tasks in industry *i* allocated to workers of type g, and  $\mathcal{T}_{ki}$  denotes those allocated to capital. We define industry-level task shares,  $\Gamma_{gi}$  and  $\Gamma_{ki}$ , as:

$$\Gamma_{gi}(\boldsymbol{w}, \Psi) = \frac{1}{M_i} \int_{\mathcal{T}_{gi}} \psi_g(x)^{\lambda - 1} \cdot dx \quad \text{and} \quad \Gamma_{ki}(\boldsymbol{w}, \Psi) = \frac{1}{M_i} \int_{\mathcal{T}_{ki}} (\psi_k(x) \cdot q(x))^{\lambda - 1} \cdot dx.$$

We assume that industry outputs are combined into a single final good (aggregate output) using a constant returns to scale aggregator,  $H(y_1, ..., y_I)$ . In the text, we work with the implied expenditure shares,  $s_i^Y(p)$ , where  $p = (p_1, p_2, ..., p_I)$  is the vector of industry prices.<sup>11</sup>

The next proposition generalizes Proposition 1 to this environment and characterizes the equilibrium in terms of task shares. As before, we denote the direct impacts of productivity deepening and automation on task shares in industry *i* by  $d \ln \Gamma_{gi}^{deep}$  and  $d \ln \Gamma_{gi}^{auto}$ , respectively.

PROPOSITION 3—Equilibrium in multi-sector economy: There is a unique equilibrium. In this equilibrium, output, wages, and industry prices can be expressed as functions of task shares defined implicitly by the solution to the system of equations:

$$w_{g} = \left(\frac{y}{\ell_{g}}\right)^{\frac{1}{\lambda}} \cdot A_{g}^{\frac{\lambda-1}{\lambda}} \cdot \left(\sum_{i \in \mathcal{I}} s_{i}^{Y}(\boldsymbol{p}) \cdot (A_{i}p_{i})^{\lambda-1} \cdot \Gamma_{gi}\right)^{\frac{1}{\lambda}},$$
(10)

$$p_i = \frac{1}{A_i} \left( A_k^{\lambda - 1} \cdot \Gamma_{ki} + \sum_{g \in \mathcal{G}} w_g^{1 - \lambda} \cdot A_g^{\lambda - 1} \cdot \Gamma_{gi} \right)^{\frac{1}{1 - \lambda}},\tag{11}$$

$$1 = \sum_{i \in \mathcal{I}} s_i^Y(\boldsymbol{p}). \tag{12}$$

The proposition shows that task shares, the  $\Gamma_{ki}$ 's and  $\Gamma_{gi}$ 's, continue to be key determinants of real wages, and we can express the equilibrium of the economy as a function of task shares, though we no longer have a closed-form solution for output. In addition, the impact of automation technologies on equilibrium outcomes again works via their influence on task shares.

<sup>&</sup>lt;sup>11</sup>For example, if *H* is CES with elasticity  $\eta$ , then  $s_i^Y(\mathbf{p}) = \alpha_i \cdot p_i^{1-\eta}$ . This formulation imposes homotheticity, which can be relaxed by allowing expenditure shares to additionally depend on the level of consumption.

#### 2.3. Wage Equation Without Ripple Effects

Under Assumption 1, the impact of a change in technology on wages can be written as

$$d\ln w_g = \frac{1}{\lambda} d\ln y + \frac{1}{\lambda} \sum_{i \in \mathcal{I}} \omega_g^i \cdot d\ln \zeta_i + \frac{\lambda - 1}{\lambda} d\ln \tilde{A}_g - \frac{1}{\lambda} \sum_{i \in \mathcal{I}} \omega_g^i \cdot d\ln \Gamma_{gi}^{\text{auto}}, \quad (13)$$

where  $d \ln \tilde{A}_g = d \ln A_g + \sum_{i \in \mathcal{I}} \omega_g^i \cdot d \ln \Gamma_{gi}^{\text{deep}}$ ,  $d \ln \zeta_i = d \ln s_i^Y + (1 - \lambda) \cdot (d \ln p_i + d \ln A_i)$ , and  $\omega_g^i$  denotes the share of group g's income earned in industry *i*, so that  $\sum_{i \in \mathcal{I}} \omega_g^i = 1$ .

Equation (13) generalizes (6) to a multi-sector economy. As before, a common productivity effect increases wages. In the presence of multiple sectors, wages additionally depend on changes in industry composition that take place in response to technological shifts. The implications of these industry changes are captured by workers' exposure to industry shifters,  $d \ln \zeta_i$  (the second term). Most centrally, group g's wage again depends on its direct task displacement—the automation-induced displacement it experiences, but now summed across all industries.

Equation (13) summarizes the key empirical prediction of our model: groups experiencing greater (automation-driven) task displacement should see *relative* wage declines. In what follows, we use this equation, which focuses on the direct effects of automation, as the basis for our reduced-form analysis. Our general equilibrium exploration in Section 5 will allow for additional ripple effects and will incorporate the wage impacts of productivity increases and induced changes in industry composition.

#### 2.4. Mapping the Model to Data and Measuring Task Displacement

Our reduced-form analysis estimates an empirical analogue of equation (13), relating wage changes of different worker groups to their task displacement. In this equation:

The common expansion of output,  $d \ln y$ , will be absorbed by the constant term.

The industry shifters term  $\sum_{i \in \mathcal{I}} \omega_g^i \cdot d \ln \zeta_i$  will be parameterized by group g's exposure to changes in industry (log) value added shares.

The third term,  $d \ln A_g$ , which incorporates factor-augmenting and productivitydeepening technologies, will be parameterized as in the SBTC literature. In particular, we assume that these technologies augment skills associated with education and gender, and impose

$$\frac{\lambda - 1}{\lambda} d \ln \tilde{A}_g = \alpha_{\text{edu}(g)} + \gamma_{\text{gender}(g)} + v_g,$$

where  $v_g$  is an additional unobserved component, and  $\alpha_{edu(g)}$  and  $\gamma_{gender(g)}$  will be absorbed by dummies for education levels and gender. As a further refinement, we allow groupspecific shifters to also depend on baseline group wages, which can be thought to proxy for skills.

■ Finally, the key explanatory variable is our measure of direct task displacement driven by advances in automation technologies,  $\sum_{i \in \mathcal{I}} \omega_g^i \cdot d \ln \Gamma_{gi}^{auto}$ .

We use two complementary strategies to measure automation-driven task displacement, both of which rely on an initial observation: displacement takes place in tasks that can be automated, which we proxy with routine tasks.<sup>12</sup> Formally, we impose:

<sup>&</sup>lt;sup>12</sup>The idea that routine tasks are easier to automate is the main premise of Autor, Levy, and Murnane (2003) and is in line with several studies that document a decline in routine jobs following automation, including Ace-

ASSUMPTION 2: Only routine tasks can be automated and, within an industry, different groups of workers are displaced from their routine tasks at a common rate.

In our reduced-form analysis, we focus on the case with  $\lambda = 1$ , which yields measures that are more transparent and easier to interpret. Appendix A-3 of the Supplemental Material shows that when  $\lambda = 1$  and Assumptions 1 and 2 hold, (direct) task displacement can be measured as

Task displacement<sup>direct</sup><sub>g</sub> = 
$$\sum_{i \in \mathcal{I}} \omega_g^i \cdot \frac{\omega_{gi}^R}{\omega_i^R} \cdot (-d \ln s_i^{L, \text{auto}}).$$
 (14)

This measure comprises three terms: (1) the first,  $\omega_g^i$ , is the share of wages earned by group g workers in industry i (relative to their total earnings) and captures this group's exposure to industry i; (2) the second term,  $\omega_{gi}^R / \omega_{gi}^R$ , parameterizes the specialization of group g in routine jobs within industry i, which are the ones directly impacted by automation. This term is computed as the share of wages earned in routine jobs in industry i by workers in group g (relative to their total earnings in that industry),  $\omega_{gi}^R$ , divided by the share of wages earned in routine jobs by all workers in industry i (relative to the total wage bill of the industry),  $\omega_i^R$ ; (3) the (percent) decline in industry i's labor share driven by automation,  $-d \ln s_i^{L,auto}$ . The automation-driven labor share decline quantifies the direct losses of routine tasks experienced by workers in an industry.

Our two measures of task displacement differ in how they treat this last term. Our first and simpler strategy assumes that the observed decline in the labor share of an industry,  $-d \ln s_i^L$ , can be entirely attributed to automation. This strategy is valid when  $\lambda = 1$  (so that factor prices do not affect the labor share), there are no changes in markups, and there are no other influences on industry wages (such as changes in worker rents). We explore later the role of these factors and provide alternative measures of task displacement that adjust for each of them.

Our second and preferred approach uses data on the adoption of automation technologies at the industry level to isolate automation-driven declines in industry labor shares. Specifically, we estimate  $-d \ln s_i^{L,\text{auto}}$  as the predicted change in the (log) labor share of an industry based on its adoption of automation technologies (and offshoring).

We present results using both strategies throughout the paper. Our preference for the second strategy is rooted in the fact that it exploits actual measures of automation, such as adoption of industrial robots, dedicated machinery, and specialized software. It also allows us to estimate the extent of task displacement generated by automation and offshoring. In our robustness checks and general equilibrium analysis, we use more general measures of task displacement that are valid when  $\lambda \neq 1$  and Assumption 1 is relaxed.

## 3. DATA, MEASUREMENT, AND DESCRIPTIVE PATTERNS

In this section, we describe our data sources and measures of task displacement, and provide a first look at the relationship between task displacement and real wage changes.

moglu and Restrepo (2020) and Humlum (2020). We present very similar results using various other measures of tasks that can be automated. Although which tasks can be automated will likely change with advances in AI, AI technologies are not present for most of our sample; Acemoglu, Autor, Hazell, and Restrepo (2022) showed that AI use takes off in the United States after 2015.

#### 3.1. Main Data Sources

We use data from the BEA Integrated Industry-Level Production Accounts on industry labor shares, factor prices, and value added for 49 industries from 1987 to 2016.<sup>13</sup> We complement these data with three industry-level proxies for adoption of automation technologies. These are: (1) change in the value of dedicated machinery services in value added between 1987 and 2016; (2) change in the value of specialized software services in value added between 1987 and 2016;<sup>14</sup> (3) the adjusted penetration of robots from 1993 to 2014, which measures robot adoption driven by international advances in technology (from Acemoglu and Restrepo (2020)). We regress changes in industry (log) labor shares between 1987 and 2016 on these three proxies for automation technologies and compute the automation-driven decline in the labor share as the predicted value in this regression.<sup>15</sup> In addition, we look at a measure of changes in intermediate imports to proxy for offshoring (from Feenstra and Hanson (1999)). Finally, to control for other trends affecting industries, we use data on total capital to value added ratio and industry TFP, sales concentration, estimates of markups, unionization rates, and measures of Chinese import competition (from Autor, Dorn, and Hanson (2013)).

On the worker side, we use Census and American Community Survey (ACS) data to trace the labor market outcomes of 500 demographic groups defined by gender, education (less than high school, high-school graduate, some college, college degree, and post-graduate degree), age (using 10-year age bins, from 16–25 years to 56–65), race/ethnicity (White, Black, Asian, Hispanic, Other), and native versus foreign-born. For each demo-graphic group, we measure real hourly wages and other labor market outcomes in 1980 (using the 1980 U.S. Census) and in 2016 (pooling data from the 2014–2018 ACS), and compute the change in real wages, employment, and non-participation rates between 1980 and 2016. In Section 4.6, we zero in on variation in labor market outcomes for demo-graphic groups across U.S. regions and commuting zones.

#### 3.2. Changes in Industry Labor Shares and Automation

Figure 3 depicts the industry-level variation in labor share declines (the basis of our first measure of task displacement, shown with the light gray/blue bars) and the component of the labor share decline driven by automation technologies (a summary measure of overall automation in the industry and the basis of our preferred task displacement variable, shown with the dark gray/orange bars) from 1987 to 2016.<sup>16</sup> The figure reveals considerable variation in industry labor share changes, with the largest declines taking place

<sup>&</sup>lt;sup>13</sup>These 49 industries can be consistently tracked in Census and BLS data, and cover the entire nongovernment sector. We have no data on our proxies of automation for the government sector. Hence, when constructing our measures of task displacement, we assume that workers in the government sector experienced no automation.

<sup>&</sup>lt;sup>14</sup>Both of these are from the BLS Total Multifactor Productivity tables. These tables also provide alternative series for labor share and factor prices in the 49 industries used in our analysis. These series are based on the same underlying data as the BEA's, but use different imputations and exclude non-profits and firms producing services that are difficult to price. All of our results are robust to using these alternative data series.

<sup>&</sup>lt;sup>15</sup>Regressing changes in labor shares on these measures is also useful for isolating the component of investments in specialized software and dedicated machinery that are related to automation (which may differ between the two types of technologies). Our exclusion restriction does not impose that all software and dedicated machinery are automation technologies, but it requires that their non-automation component is orthogonal to other factors affecting industry labor shares.

<sup>&</sup>lt;sup>16</sup>In what follows, all numbers are re-scaled to 36-year equivalent changes, so that they match the length of the time window for which we measure real wage changes (1980–2016).



FIGURE 3.—Percent decline in industry labor shares (in light gray/blue) and automation-driven labor share declines (in dark gray/orange), 1987–2016. See text for variable definitions.

in mining, chemical products, petroleum, primary metals, motor vehicles, computers and electronics, computer services, and legal services. There is also a strong correlation between the light gray (blue) and the dark gray (orange) bars, indicating that industries with the largest labor share declines are those that have been at the forefront of automation technology adoption. Industries most affected by automation are consequently similar to those listed above and include motor vehicles, primary metals, computers and electronics, computer services, plastic and rubber products, and legal services.<sup>17</sup>

Figure 4 illustrates the relationship between automation and industry labor share changes. Panel A depicts a strong negative association between labor share changes and adjusted penetration of robots ( $R^2 = 0.18$ ). Panel B shows this association for the combined change in specialized software and dedicated machinery services ( $R^2 = 0.32$ ). Panel C presents the relationship between observed labor share changes and the predicted labor share decline based on our three proxies of automation, which together account for 45% of the variation in industry labor share changes. Table A-I in the Supplemental Material further explores this relationship. It shows that offshoring matters for the labor share decline as well, but accounts for only 2% of the variation across industries.

Table A-I also confirms that the inclusion of changes in total capital to value added ratio, sales concentration, markups, import competition, and unionization rates does not change the correlation between our proxies of automation and industry labor share

<sup>&</sup>lt;sup>17</sup>Assumption 2 receives support from industry-level variation as well. Figure B-1 and Table B-2 in Appendix B-4 document a strong negative association between labor share declines, or its automation-driven component, and reductions in the demand for routine tasks across industries (measured in one of three ways: total wages in routine jobs, total hours in routine jobs, or total number of workers in routine jobs).



FIGURE 4.—Relationship between automation technologies and changes in industry labor shares. See text for variable definitions. The five industries with the highest and the five industries with the lowest changes in their labor shares are identified in the figures.

changes. In fact, conditional on our proxies of automation, these variables do not have a sizable or statistically significant effect on industry's labor shares. Note also that changes in total capital to value added ratio are a "bad control," since our proxies of automation all contribute to the capital stock. Nevertheless, the fact that this variable has no discernible effect on our results suggests that specialized software, dedicated equipment, and industrial robots capture types of capital that lead to sizable declines in labor shares, presumably because they are used for automation, while other forms of capital are not.

# 3.3. Task Displacement and Wages Across Demographic Groups

We compute (direct) task displacement for our 500 demographic groups using equation (14). Specialization patterns across industries and routine jobs, the  $\omega$  terms, are computed from the 1980 Census—a year that predates major advances in automation technologies—while  $-d \ln s_i^{L,\text{auto}}$  corresponds to the 1987–2016 change in industry labor share or its component driven by automation technologies, as described in the previous subsection.<sup>18</sup>

Figure 5 presents our two measures of task displacement for the 500 groups of workers. Panel A shows that these two measures—one computed from changes in the labor share on the horizontal axis, and the other exploiting the component driven by automation on the vertical axis—are strongly correlated ( $R^2 = 0.95$ ). This figure also reveals sizable differences in task displacement across demographic groups, using either measure: some demographic groups experienced a 25% direct reduction due to automation between 1980 and 2016, while others saw no change in their task shares.

Panel B plots our measure of task displacement based on automation-driven labor share declines for all groups sorted by their baseline wage in 1980. We see that (direct) task

1989

<sup>&</sup>lt;sup>18</sup>We created a consistent mapping of the 49 industries in the BEA data to the Census industry classification. For each industry, we computed the share of wages earned in routine jobs by a demographic group, using the definition of routine occupations described in Acemoglu and Autor (2011), where a third of the occupations in 1980 are classified as routine. Further details on the data used are provided in Appendix B-3. If instead we use data from the 2000 Census, the resulting task displacement measure is very similar (with a rank correlation of 0.93 with our baseline measure), confirming the strong persistence of specialization patterns.



FIGURE 5.—Direct task displacement measures for the 500 demographic groups in our sample. The left panel shows a scatter plot between our two task displacement measures. The first, computed from observed labor share declines, is on the horizontal axis, while the second, computed from automation-driven labor share declines, is on the vertical axis. The 45° degree line is shown in black. The right panel plots our measure of task displacement computed from automation-driven labor share declines against the baseline hourly wages of groups in 1980. Marker sizes indicate the share of hours worked by each group and shades of gray (different colors) indicate education levels. See text for variable definitions.

displacement has been particularly high during this period for groups in the middle of the wage distribution—thus playing both an unequalizing and a polarizing role.

Figure 6 provides a first glimpse of the association between (direct) task displacement and real wage changes across demographic groups. The top two panels plot the bivariate relationship between our two task displacement measures and real wage changes from 1980 to 2016. These plots reveal a strong correlation between task displacement and changes in real wages, with groups experiencing the highest levels of task displacement seeing their real wages fall or stagnate. The bottom two panels display a simple falsification exercise. They demonstrate that the relationships depicted in the top two panels are not driven by secular trends adversely affecting some groups and are not present between 1950 and 1980—a period that predates major advances in automation. Rather, all demographic groups, including those that experienced adverse task displacement after 1980, enjoyed robust real wage growth, of about 50%, between 1950 and 1980.

Both figures identify different education levels, highlighting that task displacement has been much higher for workers without a college degree. Consequently, workers without college have much lower, and in some cases negative, real wage changes. The relationship between task displacement and real wage changes is not just between education groups, however: a negative association between changes in wages and task displacement *within* education groups is visible from this figure as well.

Table A-II in the Supplemental Material provides descriptive statistics for the 500 demographic groups in our analysis and further corroborates these patterns. For example, it shows that workers in the top quintile of the (direct) task displacement distribution saw their real wage decline by 12%, while workers in the least exposed groups enjoyed real wage growth of about 26%.



FIGURE 6.—Reduced-form relationship between task displacement and changes in real hourly wages. Panel A plots changes in real hourly wages for 1980–2016 against our task displacement measure computed from observed labor share declines. The slope of the regression line is -1.6 (standard error = 0.09). Panel B plots changes in real wages for 1980–2016 against our task displacement measure computed from automation-driven labor share declines for 1980–2016. The slope of the regression line is -1.65 (standard error = 0.10). Panels C and D plot pre-trends (changes in real hourly wages for 1950–1980) against our two task displacement measures for 1980–2016. The slopes of the regression lines in both Panels C and D are -0.28 (standard error = 0.28). Marker sizes indicate the share of hours worked by each group and shades of gray (different colors) indicate education levels. See text for variable definitions.

#### 4. REDUCED-FORM EVIDENCE OF THE EFFECTS OF TASK DISPLACEMENT

This section presents our main reduced-form results. It highlights how automationinduced task displacement explains a large fraction of the changes in the U.S. wage structure between 1980 and 2016.

# 4.1. Baseline Results

Table I presents our baseline estimates from an empirical analogue of equation (13):

 $\Delta \ln w_g = \beta^d \cdot \text{Task displacement}_g^{\text{direct}} + \beta^s \cdot \text{Industry shifters}_g + \alpha_{\text{edu}(g)}$ 

1991

## TABLE I

TASK DISPLACEMENT AND CHANGES IN REAL HOURLY WAGES, 198	0-2016.
---	---------

		Dependen Change in Hourly	t Variables: Wages, 1980–2016	
	(1)	(2)	(3)	(4)
	PANEL A. TA	ASK DISPLACEMENT B	ASED ON LABOR SHAI	RE DECLINES
Task displacement	-1.60	-1.32	-1.31	-1.66
Ĩ	(0.09)	(0.16)	(0.19)	(0.44)
Industry shifters	× ,	0.21	0.31	0.35
5		(0.09)	(0.12)	(0.16)
Exposure to industry		~ /		0.18
labor share decline				
				(0.66)
Relative specialization				0.07
in routine jobs				0.07
in routine joos				(0.07)
Share variance	0.67	0.55	0.55	0.70
explained by task	0.07	0.55	0.55	0.70
displacement				
R-squared	0.67	0.70	0.84	0.84
Observations	500	500	500	500
Observations	DANEL B T	JUU ASK DISDI ACEMENT I	JUU DASED ON AUTOMATI	
	IANEL D. I	LADOD CHAI	DASED ON AUTOMATT	UN-DRIVEN
Task displacement	1.65	1 41	1 26	1.96
lask displacement	-1.03	-1.41	-1.50	-1.60
In ductory abift and	(0.10)	(0.20)	(0.21)	(0.47)
industry shifters		0.15	0.10	0.20
E		(0.11)	(0.14)	(0.16)
Exposure to industry				-0.08
labor share decline				(0,00)
				(0.80)
Relative specialization				0.10
in routine jobs				(0.00)
~ .				(0.08)
Share variance	0.64	0.55	0.53	0.72
explained by task				
displacement				
<i>R</i> -squared	0.64	0.66	0.83	0.83
Observations	500	500	500	500
Other covariates:				
Manufacturing share,			$\checkmark$	$\checkmark$
education and gender				
dummies				

*Note:* This table presents estimates of the relationship between task displacement and the change in hourly wages across 500 demographic groups, defined by gender, education, age, race, and native/immigrant status. The dependent variable is the change in hourly wages for each group between 1980 and 2016. Panel A reports results for our measure of task displacement based on observed labor share declines. Panel B reports results for our measure of task displacement based on automation-driven labor share declines. In addition to the covariates reported in the table, columns 3 and 4 control for baseline wage shares in manufacturing and dummies for education (for no high-school degree, completed high school, some college, college degree, and post-graduate degree) and gender. All regressions are weighted by total hours worked by each group in 1980. Standard errors robust to heteroskedasticity are reported in parentheses.

Here g indexes our 500 demographic groups, and  $\Delta \ln w_g$  denotes the log change in real hourly wages for workers in group g between 1980 and 2016. The error term  $v_g$  represents residual group-specific changes in supply or demand. As in all of our other results,

Our identifying assumption is that the two (direct) task displacement measures are uncorrelated with other trends affecting wages—except through automation-driven task displacement.

Panel A of the table presents results with our first measure of task displacement, constructed from observed declines in industry labor shares. Panel B presents results with our second, preferred measure of task displacement, which focuses on the component of the labor share declines driven by automation technologies.

Column 1 presents a bivariate regression identical to the one shown in Figure 6. In Panel A, we see a precise and sizable relationship between task displacement and wage growth, with a coefficient of -1.6 (s.e. = 0.09). This estimate implies that a 25% increase in task displacement—which corresponds to the displacement experienced by white American men aged 26–35 with no high-school degree—is associated with a 40% (relative) wage decline. The bottom rows report the share of wage changes explained by task displacement.<sup>19</sup> Our measure of task displacement alone explains 67% of the variation in wage changes between 1980 and 2016.

The remaining columns document that this bivariate relationship is robust. Column 2 controls for industry shifters, which absorb labor demand changes coming from the expansion of industries in which a demographic group specializes. The coefficient estimate for task displacement is similar to the one in column 1, -1.32 (s.e. = 0.16). Column 3, which we take as our baseline specification for the rest of the paper, controls for gender and education dummies and a group's share of earnings in manufacturing. These account for other demand factors favoring highly-educated workers and for the effects of the secular decline of manufacturing. The coefficient estimate remains very similar to column 2, -1.31 (s.e. = 0.19). Even after the inclusion of these controls, task displacement continues to explain 55% of the variation in wage changes during this period.

Our first task displacement measure in Panel A combines industry-level changes in labor shares with the distribution of employment of workers across industries and (routine and non-routine) occupations. Column 4 includes two more variables, corresponding to the constituent parts making up our task displacement measure. The first is the exposure of a demographic group to industry-level declines in the labor share, but without focusing on whether employment is in routine tasks in that industry. The second is a group's relative specialization in routine jobs, but this time without exploiting industry-level changes in task displacement.<sup>20</sup> Column 4 shows that these two variables themselves do not explain real wage changes (conditional on task displacement), while task displacement remains very strongly correlated with wage changes. This result confirms that our measure of task displacement is not confounded by other industry-level changes impacting labor shares

<sup>20</sup>Formally, these controls are defined as

exposure to industry labor share declines<sub>g</sub> = 
$$\sum_{i \in \mathcal{I}} \omega_g^i \cdot (-d \ln s_i^{L, \text{auto}})$$
  
relative specialization in routine jobs<sub>g</sub> =  $\sum_{i \in \mathcal{I}} \omega_g^i \cdot \frac{\omega_g^R}{\omega_i^R}$ .

<sup>&</sup>lt;sup>19</sup>Following Klenow and Rodríguez-Clare (1997), we decompose the variance of y in the linear model  $y = \sum_i x_i \beta_i + \varepsilon$  as  $\operatorname{Var}(y) = \sum_i \beta_i \cdot \operatorname{Cov}(y, x_i) + \operatorname{Cov}(y, \varepsilon)$  and compute the share of the variance in y explained by  $x_i$  as  $\beta_i \cdot \frac{\operatorname{Cov}(y, x_i)}{\operatorname{Var}(y)}$ . These shares add up to the  $R^2$  of the regression, which is also reported in the table.

and wages or by other trends affecting workers specializing in routine tasks. Rather, it is demographic groups specializing in routine tasks in industries undergoing sizable labor share declines that suffer relative wage declines. The lack of a negative impact on groups specializing in non-routine tasks also confirms that our results are not driven by a mechanical association between changes in the average wages paid in an industry and changes in its labor share.

Panel B presents results using our preferred measure of task displacement based on the component of the labor share decline driven by automation technologies. The estimates of the effects of task displacement and the shares of variance explained by this variable are, in all cases, very similar to those in Panel A. In column 3, for example, the coefficient estimate of task displacement is -1.36 (standard error = 0.21), compared to -1.31 in the same specification in Panel A. The share of wage structure changes explained by task displacement in this column is also similar: 53%, compared to 55% in Panel A.<sup>21</sup>

Overall, our baseline results document a strong association between task displacement and real wage changes, and support our presumption that this association is capturing the (direct) causal effect of automation-driven task displacement experienced by a demographic group. We next bolster the case that this is indeed a causal relationship and is highly robust.

# 4.2. Instrumental-Variables Estimates

Our preferred measure of task displacement is based on the component of the industry labor share decline that is driven by our proxies of automation. A complementary approach entails using these proxies as instruments for our first measure of task displacement based on observed labor share declines. Table II pursues this approach and confirms that estimating equation (15) via two-stage least squares (2SLS) yields similar results.<sup>22</sup>

Panel A presents 2SLS estimates for a specification analogous to column 3 of Table I. Column 1 uses all three of our proxies as instruments. The first-stage *F*-statistic is very high (846.9). The 2SLS estimate for the effect of task displacement, -1.23 (s = 0.19), and the implied share of variance of wage changes explained by task displacement, 50%, are similar to those in column 3 of Table I, obtained by directly using our second measure of task displacement.

The remaining columns explore the contribution of each of the proxies of automation. Column 2 uses the adjusted penetration of robots by itself. Columns 3 and 4 focus on dedicated machinery and specialized software as proxies for automation. Columns 5 and 6 include the software measure together with each one of the other two proxies for automation. The 2SLS estimates are similar across columns 1–6 and the hypothesis that they

<sup>&</sup>lt;sup>21</sup>Our main tables report robust standard errors. More conservative standard errors (for our baseline specification in column 3) are presented in Table B-3 in the Supplemental Material and confirm our main findings. In particular, we compute standard errors as in Adao, Kolesár, and Morales (2019) and Borusyak, Hull, and Jaravel (2022), which are robust to the presence of unobserved industry shocks that affect all workers or workers in routine jobs in an industry. We additionally report standard errors from the single-step GMM estimation of the predicted decline in the labor share and our wage equation (see Newey (1984)), which corrects for the fact that the task displacement measure in Panel B is itself estimated from an industry-level regression.

<sup>&</sup>lt;sup>22</sup>Formally, we use instruments of the form  $\sum_{i \in \mathcal{I}} \omega_g^i \cdot \frac{\omega_g^{i'}}{\omega_e^R} \cdot Automation proxy_i$ , where our proxy is the adjusted penetration of robots, changes in the value of dedicated machinery services in value added, or changes in the value of specialized software services in value added, or combinations thereof. When we include exposure to industry labor share declines as a covariate in Panel B, we use variables of the form  $\sum_{i \in \mathcal{I}} \omega_g^i \cdot Automation proxy_i$  as instruments.

TABLE II	SLS ESTIMATES USING AUTOMATION AND OFFSHORING AS INSTRUMENTS.
----------	---

		Depend	ent Variable: Cha	nge in Hourly Wag	es 1980–2016		
	Robot APR,				Robot	Machinery	
	Machinery, and		Dedicated	Specialized	APR and	and	
Instruments:	Software	Robot APR	Machinery	Software	Software	Software	Offshoring
	(1)	(2)	(3)	(4)	(5)	(9)	(1)
	PANEL A. 2SLS EST	<b>TIMATES INSTRUMEN</b>	VTING TASK DISPL	ACEMENT WITH OU	JR AUTOMATION	AND OFFSHORING	PROXIES
Task displacement	-1.23	-1.22	-0.83	-1.46	-1.33	-1.20	-0.81
	(0.19)	(0.25)	(0.39)	(0.36)	(0.21)	(0.18)	(0.30)
Share variance explained	0.50	0.39	0.17	0.22	0.52	0.49	0.12
by task displacement							
R-squared	0.84	0.84	0.83	0.84	0.84	0.84	0.82
First-stage F	846.91	98.00	29.53	68.00	432.89	716.72	30.62
Overid <i>p</i> -value	0.07				0.58	0.33	
Observations	500	500	500	500	500	500	500
	PANEL ]	<b>B. 2SLS ESTIMATES</b>	CONTROLLING FC	R INDUSTRY AND	OCCUPATIONAL S	<b>PECIALIZATION</b>	
Task displacement	-1.56	-1.26	-0.08	-3.11	-2.06	-1.36	-2.49
ſ	(0.50)	(0.83)	(0.97)	(1.12)	(0.56)	(0.49)	(0.71)
Exposure to industry labor	0.54	0.30	-1.91	-1.19	0.15	0.30	0.06
share decline							
	(0.76)	(0.91)	(1.57)	(1.63)	(0.85)	(0.77)	(0.96)
Relative specialization in	0.06	0.01	-0.18	0.29	0.13	0.03	0.20
routine jobs							
	(0.0)	(0.14)	(0.17)	(0.17)	(0.09)	(0.0)	(0.11)
Share variance explained	0.63	0.41	0.02	0.47	0.80	0.55	0.37
by task displacement							
R-squared	0.84	0.83	0.82	0.76	0.84	0.84	0.83
First-stage F	170.40	6.32	30.15	3.87	26.12	190.09	23.71
Observations	500	500	500	500	500	500	500
<i>Note:</i> This table presents 2SLS est native/immigrant status. The dependent to instrument for our measure of task of	imates of the relationship betw t variable is the change in hourly displacement based on observec	een task displacement a wages for each group b 1 labor share declines. F	nd changes in hourly etween 1980 and 2016 ormally, we use instri	wages for 500 demogring the form $\sum_{n=1}^{\infty}$ such that the form $\sum_{n=1}^{\infty}$	aphic groups, define estimates using our $i \in \mathcal{I} \ \omega^i_g \cdot (\omega^R_{gi}/\omega^R_i) \cdot$	ed by gender, education measures of automation Automation proxy <sub>i</sub> , w	on, age, race, and on and offshoring there our proxy is

TASKS, AUTOMATION, AND RISE IN U.S. WAGE INEQUALITY

also control for relative specialization in routine jobs and exposure to industry labor share declines (this last term instrumented also using our proxies for technology and offshoring). Formally, the models in this panel also use instruments of the form  $\sum_{i \in \mathcal{I}} \sigma_i^g$ . Automation proxy, in addition to the covariates reported in the table, all specifications control for industry shifters, group's baseline either the adjusted penetration of robots, our measures of changes in dedicated machinery and specialized software services, or our measure of offshoring. Panel B provides 2SLS estimates where we

wage share in manufacturing, and dummies for education (for no high-school degree, completed high school, some college, college degree, and post-graduate degree) and gender. All regressions are

weighted by total hours worked by each group in 1980. Standard errors robust to heteroskedasticity are reported in parentheses.

are all equal cannot be rejected. This finding is consistent with our exclusion restriction that the effects of these technologies operate through task displacement.

Finally, in column 7, we turn to offshoring—measured as the change in the share of imported intermediates in an industry. As expected, offshoring also contributes to task displacement and depresses real wages of exposed groups, but it only explains 12% of the variation in wage changes.

Panel B of the table presents estimates corresponding to the specification in column 4 of Table I (thus also controlling for exposure to industry labor share declines, instrumented by our automation proxies, and for relative specialization in routine jobs). The results in this panel are similar, though less precise in some specifications (such as in columns 2 and 3 where we use, respectively, the adjusted penetration of robots and dedicated machinery as the only instruments).

In what follows, we focus on estimates using our measure of task displacement based on automation-driven labor share declines, as in Panel B of Table I, rather than the 2SLS models presented in this subsection.

### 4.3. Task Displacement versus SBTC

How important is task displacement relative to other forms of SBTC? Table III explores this question by considering different specifications of SBTC. The first column of this table regresses wage changes on a full set of dummies for gender and education levels, but excludes our task displacement measure. As explained in Section 2.4, these controls absorb any factor-augmenting productivity trends common to all workers with the same education level or gender. Column 1 shows that these SBTC variables are significant and have the expected signs. For example, between 1980 and 2016, the relative wage of workers with a college (but no post-graduate) degree increased by 25% relative to those with a high-school degree, and the relative wage of workers with a post-graduate degree increased by 42% relative to high-school graduates. In this model, education dummies explain 55% of the variation in wage changes during this period.

However, most of the differences between workers with different education levels disappear once our task displacement measures are included in columns 2 and 3 (these models are identical to column 3 in Panels A and B of Table I). Notably, there is no longer any differential wage growth for workers with a college degree relative to those with a highschool degree. Likewise, task displacement explains, respectively, 80% and 65% of the rise in the post-graduate premium with our two measures. Task displacement also explains more than 50% of the overall changes in the wage structure, while the education dummies explain less than 17%. In addition, task displacement accounts for 4–7 percentage points of the 17% decline in the gender wage premium during this period. These results are the basis for our claim that much of the change in the U.S. wage structure between 1980 and 2016 is due to task displacement, with a minor role for standard (factor-augmenting) SBTC.

The next three columns go one step further and allow for differential trends that depend on the baseline wage level of each demographic group, which could proxy for dimensions of group skills that go beyond education and gender. The results of these demanding specifications are similar to those in the first three columns, and our task displacement measure explains about 40% of the observed wage changes, while differential trends by education and baseline wages explain, respectively, 18% and 7% of the variation.

In Table A-III in the Supplemental Material, we also control for the differential evolution of the supply (population size) of different demographic groups, which is the equivalent of the relative supply controls in Katz and Murphy (1992) and Card and Lemieux

			Dependent Variable: Change	in Hourly Wag	ces 1980–2016	
		SBTC by Education Le	evel and Gender		SBTC by Wage	e Level
		Labor Share	Automation-Driven		Labor Share	Automation-Driven
Task Displacement Measure		Declines	Declines		Declines	Declines
	(1)	(2)	(3)	(4)	(5)	(9)
Gender: women	0.17	0.10	0.13	0.25	0.15	0.17
	(0.02)	(0.02)	(0.02)	(0.02)	(0.03)	(0.03)
Education: no high school	0.02	0.02	0.03	0.05	0.04	0.05
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Education: some college	0.05	-0.07	-0.05	0.03	-0.06	-0.03
	(0.03)	(0.03)	(0.03)	(0.02)	(0.03)	(0.03)
Education: full college	0.25	-0.02	0.03	0.18	0.01	0.05
	(0.04)	(0.05)	(0.05)	(0.04)	(0.05)	(0.05)
Education: more than college	0.42	0.08	0.15	0.29	0.09	0.16
	(0.05)	(0.06)	(0.06)	(0.05)	(0.06)	(0.06)
Log of hourly wage in 1980				0.23	0.12	0.11
				(0.05)	(0.04)	(0.05)
Task displacement		-1.31	-1.36		-1.03	-1.01
		(0.19)	(0.21)		(0.18)	(0.24)
Share variance explained by:						
	0.55	0.08	0.17	0.37	0.09	0.18
-baseline wage				0.15	0.07	0.07
<ul> <li>—task displacement</li> </ul>		0.55	0.53		0.43	0.39
R-squared	0.76	0.84	0.83	0.81	0.85	0.83
Observations	500	500	500	500	500	500
Other covariates:						
Industry shifters and	>	>	>	>	>	>
manufacturing share						

TABLE III Task displacement versus SBTC, 1980–2016. Note: This table presents estimates of the relationship between SBTC proxies, task displacement, and the change in hourly wages across 500 demographic groups. These groups are defined by gender, education, age, race, and native/immigrant status. The dependent variable is the change in hourly wages for each group between 1980 and 2016. Columns 2 and 5 report results using our measure of task displacement based on observed labor share declines. Columns 3 and 6 report results using our measure of task displacement based on automation-driven labor share declines. In addition to the covariates reported in the table, all specifications control for industry shifters and baseline wage shares in manufacturing. The bottom rows of the table report the share of variance explained by task displacement and the different proxies of skill-biased technical change. All regressions are weighted by total hours worked by each group in 1980. Standard errors robust to

heteroskedasticity are reported in parentheses.

#### TASKS, AUTOMATION, AND RISE IN U.S. WAGE INEQUALITY

(2001). The inclusion of these controls raises the explanatory power of our task displacement measure (because demographic shifts have gone in favor of groups experiencing task displacement). Now, task displacement explains 63%–72% of the changes in the U.S. wage structure, while the education dummies continue to explain a small portion (4%–18%) of the variation.

In summary, our results show that task displacement has been at the root of the changes in the wage structure from 1980 to today, while other forms of SBTC had limited explanatory power.

#### 4.4. Employment Outcomes

If task displacement leads to lower labor demand for a demographic group, we should see an impact not just on its wage but on its employment as well.<sup>23</sup> Table IV presents results for the employment to population ratio in the top panel and non-participation rate in the bottom panel.

We find that task displacement is associated with lower employment to population ratios. The first three columns use the measure of task displacement based on industry labor share declines, while the next three columns rely on our preferred measure, exploiting the component of labor share declines driven by automation technologies. Panel B reveals that most of the adjustment takes place via non-participation. For example, using the estimates from column 5 based on our preferred measure, we see that a 10 percentage point higher task displacement is associated with a 4.4 percentage point decline in employment between 1980 and 2016, and a similar 3.5 percentage point increase in non-participation. Additionally, columns 3 and 6 in both panels confirm that the employment effects do not reflect adverse trends against all workers specialized in routine jobs or those employed in industries with declining labor shares. Rather, as with our wage results, they are driven by task displacement. Overall, our task displacement measure explains between 16% and 38% of changes in employment and participation between 1980 and 2016.<sup>24</sup>

# 4.5. Confounding Trends: Capital, TFP, Deunionization, Imports, and Markups

The main challenge in interpreting our reduced-form estimates as the causal effect of automation on relative wages is the possibility that labor share changes or their component driven by automation technologies are confounded by other industry-level trends. We directly confront these threats to identification in this subsection.

A first concern is that our task displacement measures may capture not just the effects of automation but of other investments or other types of technological change. We already saw in Table A-I that, conditional on our automation measures, changes in the total capital to value added ratio are not correlated with industry labor shares. Columns 1–2 and 5–6 in Panel A of Table V further explore the role of capital intensity and industry productivity by controlling for exposure to changes in industry capital to value added ratios and TFP growth. Consistent with our interpretation that wage inequality is shaped by task displacement, rather than overall capital intensity or productivity growth resulting

<sup>&</sup>lt;sup>23</sup>Appendix B-1 provides an extension of our model that allows for endogenous supply responses and shows that task displacement will lead to a relative decline in hours worked. This decline could be involuntary if the labor market is not competitive (see, e.g., Kim and Vogel (2021)).

<sup>&</sup>lt;sup>24</sup>We report results for hours per worker and the unemployment rate in Appendix Table B-4. The responses at these margins are smaller and less robust than those of employment to population ratio and non-participation.

		Depen	dent Variable: Labor	Market Outcomes 198	30-2016	
	Task Disp	acement Based on La	abor Share	Task Displace	ment Based on Automa	ttion-Driven
		Declines			Labor Share Declines	
	(1)	(2)	(3)	(4)	(5)	(9)
		PA	NEL A. EMPLOYMEN	T TO POPULATION RAT	OL	
Task displacement	-0.68	-0.46	-0.78	-0.75	-0.44	-0.82
	(0.11)	(0.14)	(0.32)	(0.11)	(0.16)	(0.39)
Share variance explained by:						
	0.31	0.21	0.36	0.35	0.20	0.38
educational dummies		0.10	0.12		0.15	0.15
R-squared	0.31	0.77	0.78	0.35	0.77	0.78
Observations	500	500	500	500	500	500
			PANEL B. NON-PA	ARTICIPATION RATE		
Task displacement	0.67	0.37	0.77	0.75	0.35	0.80
4	(0.12)	(0.14)	(0.31)	(0.12)	(0.16)	(0.39)
Share variance explained by:			~	~	~	,
	0.30	0.17	0.34	0.33	0.16	0.36
		0.16	0.19		0.21	0.22
<i>R</i> -squared	0.30	0.80	0.81	0.33	0.80	0.81
Observations	500	500	500	500	500	500
Covariates:						
Industry shifters, manufacturing		>	>		>	>
share, education, and gender						
dummies						
Exposure to labor share declines			>			>
and relative specialization in						
routine jobs						
Note: This table presents estimates of the re-	elationship between task d	isplacement and labor ma	arket outcomes for 500 d	emographic groups, define	d by gender, education, age	, race, and na-

TASK DISPLACEMENT AND EMPLOYMENT OUTCOMES, 1980–2016.

TABLE IV

measure of task displacement based on automation-driven labor share declines. In addition to the covariates reported in the table, columns 2–3 and 5–6 control for industry shifters, baseline wage shares in manufacturing, and dummies for education (for no high-school degree, completed high school, some college, college degree, and post-graduate degree) and gender. Columns 3 and 6 control for relative specialization in routine jobs and groups' exposure to industry labor share declines. All regressions are weighted by total hours worked by each group in 1980. Standard errors robust to twe/immigrant status. In Panel A, the dependent variable is the change in the employment to population ratio between 1980 and 2016. In Panel B, the dependent variable is the change in the non-participation rate between 1980 and 2016. Columns 1-3 report results using our measure of task displacement based on observed labor share declines. Columns 4-6 report results using our heteroskedasticity are reported in parentheses.

#### TASKS, AUTOMATION, AND RISE IN U.S. WAGE INEQUALITY

1999

$\sim$	-
-	
ΓI	ļ
1	
μ	1
<	٢
r Ì	٦
	ARIF V

Task displacement and changes in real hourly wages—controlling for other trends, 1980–2016.

			mannadari	VallaUL. VIIal	Task Displacemer	nt Based on Autor	mation-Driven La	bor Share
	Task Displa	cement Based on	Labor Share Decl	ines		Decline	Ş	
			Change in				Change in	
Other shocks:	Changes in $K/Y$ Ratio by Industry	Changes in TFP by Industry	Chinese Import Competition	De-Unionizatic Rates	on Changes in $K/Y$ Ratio by Industry	Changes in TFP by Industry	Chinese Import Competition	De-Unionization Rates
	(1)	(7)	(3)	(4)	(c)	(0)	(/)	(8)
			PANEL A. CON	VTROLLING FOR	. MAIN EFFECT OF OT	THER SHOCKS		
Task displacement	-1.31	-1.31	-1.26	-1.31	-1.36	-1.38	-1.28	-1.32
ſ	(0.17)	(0.19)	(0.20)	(0.22)	(0.20)	(0.22)	(0.22)	(0.23)
Exposure to industry shock	0.01	-0.04	0.01	0.02	0.01	$-0.11^{\circ}$	0.02	-1.08
	(0.13)	(0.37)	(0.01)	(0.84)	(0.14)	(0.37)	(0.01)	(0.77)
Share variance explained by:								
-task displacement	0.55	0.55	0.53	0.55	0.53	0.54	0.50	0.51
	0.00	0.00	-0.01	-0.00	0.00	0.01	-0.02	0.16
R-squared	0.84	0.84	0.84	0.84	0.83	0.83	0.83	0.83
Observations	500	500	500	500	500	500	500	500
			PANEL B. CONTRO	<b>JLLING FOR EFI</b>	FECTS ON WORKERS I	N ROUTINE JOBS		
Task displacement	-0.87	-1.30	-1.14	-1.64	-0.75	-1.46	-1.25	-2.20
a.	(0.22)	(0.21)	(0.27)	(0.45)	(0.27)	(0.26)	(0.35)	(0.67)
Exposure to industry shock	0.28	-0.02	0.03	-0.32	0.36	-0.27	0.03	-2.42
	(0.18)	(0.46)	(0.03)	(0.76)	(0.18)	(0.49)	(0.03)	(0.89)
Exposure of routine jobs to industry shock	-0.28	-0.03	-0.01	0.74	-0.37	0.15	-0.00	1.71
	(0.15)	(0.21)	(0.02)	(0.82)	(0.15)	(0.23)	(0.02)	(1.18)
Share variance explained by:								
—task displacement	0.36	0.55	0.48	0.69	0.29	0.57	0.49	0.86
	0.12	0.01	0.02	-0.16	0.15	-0.02	-0.02	-0.11
R-squared	0.84	0.84	0.84	0.84	0.84	0.83	0.83	0.83
Observations	500	500	500	500	500	500	500	500
<i>Note:</i> This table presents est unionization rates, other forms	imates of the relation of capital investments	ship between task d s, and other technol	isplacement and the ogies leading to pro	change in hourly ductivity growth	wages across 500 demog in an industry. These gi	graphic groups controups are defined b	rolling for trade in fi by gender, education	inal goods, declining 1, age, race, and na-
	-		· ·					

on observed labor share declines. Columns 5-8 report results using our measure of task displacement based on automation-driven labor share declines. In addition to the covariates reported in the industries. In Panel B, we allow these shocks to have a differential impact on workers in routine jobs in exposed industries. Columns 1-4 report results using our measure of task displacement based table, all specifications control for industry shifters, baseline wage share in manufacturing, and dummies for education (for no high-school degree, completed high school, some college, college degree tive/immigrant status. The dependent variable is the change in hourly wages for each group between 1980 and 2016. In Panel A, we control for the main effect of these shocks on workers in exposed and post-graduate degree) and gender. All regressions are weighted by total hours worked by each group in 1980. Standard errors robust to heteroskedasticity are reported in parentheses.

## 2000

#### D. ACEMOGLU AND P. RESTREPO

from non-automation technologies, exposure to these industry variables has no direct impact on wages and does not alter the relationship between task displacement and wages.<sup>25</sup> These results support our conceptual framework and identifying assumption.

In Appendix Table A-IV of the Supplemental Material, we adopt a complementary strategy to control for changes in capital intensity. We relax both Assumption 1 and  $\lambda = 1$ , so that the elasticity of substitution between capital and labor is no longer equal to 1. We then adjust our task displacement measures for changes in industry capital utilization and other movements in factor prices. The results of this exercise are similar to our baseline findings.<sup>26</sup>

Changes in worker bargaining power can also impact industry labor shares and worker wages. To investigate this issue, in columns 3 and 7, we include workers' exposure to industries with declining unionization rates, which may have reduced their rents, thus contributing both to labor share declines and changes in the wage structure. Finally, columns 4 and 8 include the exposure of different demographic groups to industries facing greater Chinese import competition. Although industry shifters already account for the effects of trade in final goods, this specification controls for other effects of trade with China, such as those working through changes in rent-sharing. With both controls, our task displacement measures have similar coefficients to the ones we saw in Table I and continue to explain about 50% of the changes in the U.S. wage structure. We find no evidence that declining unionization rates or Chinese import competition has a direct impact on the wage structure (beyond their potential effects working through changes in industry composition).

Panel B of the table shows similar results when we allow each of these industry shocks to have a differential impact on workers specializing in routine jobs. In all cases, the effects of task displacement on wages are largely unaffected.<sup>27</sup>

Another important concern centers on the role of changes in industry concentration and markups, which also impact industry labor shares and might directly affect the wage structure. Table VI explores the role of these factors. Columns 1 and 5 in Panel A control for workers' exposure to industries with rising sales concentration, and the other columns include their exposure to markup changes using three alternative estimates of industry markups. These are: markups computed from accounting data, markups estimated using the (inverse of the) material share, and markup estimates following the production function approach in De Loecker, Eeckhout Jan, and Unger (2020) (see Appendix B-3 for details).

<sup>26</sup>When  $\lambda \neq 1$  and there are ripple effects, our baseline task displacement measure in equation (14) becomes

Task displacement<sup>direct</sup><sub>g</sub> = 
$$\sum_{i \in \mathcal{I}} \omega_g^i \cdot \frac{\omega_{g^i}^R}{\omega_i^R} \cdot \frac{-d\ln s_i^L - s_i^K \cdot (1 - \sigma_i) \cdot (d\ln \omega_i - d\ln R_i)}{1 + (\lambda - 1) \cdot s_i^L \cdot \pi_i}$$

where the adjusted labor share decline,  $-d \ln s_i^L - s_i^K \cdot (1 - \sigma_i) \cdot (d \ln w_i - d \ln R_i)$ , accounts for the influence of factor prices. In this expression,  $\sigma_i$  is the elasticity of substitution between capital and labor in industry *i*, and  $d \ln w_i$  and  $d \ln R_i$  are the observed change in factor prices faced by the industry. Table A-IV provides results using both this adjusted measure and its analogue focusing on automation-driven labor share declines.

<sup>27</sup>Table B-5 shows that the results are similar when we control for exposure to labor share declines and relative specialization in routine occupations as in column 4 of Table I.

<sup>&</sup>lt;sup>25</sup>These results are in line with our theoretical expectations. For example, when Assumption 1 holds, higher productivity of capital in tasks it is already performing will lead to greater capital utilization but will not cause any task displacement, and as such will not negatively impact any workers. Likewise, industry TFP growth should only affect labor demand through the industry shifters, which are already being controlled for in these regressions.

Task displacement and changes in real hourly wages—controlling for changes in markups and industry concentration, 1980–2016.

			Dependent	Variable: Change	in Hourly Wages 19	980-2016		
	Task Displacen	nent Based on La	bor Share Declines		Task Displaceme	ant Based on Auto Share Decline	mation-Driven Lab ss	oor
		Markups From		- -		Markups From	- -	- -
	Change in Sales Concentration (1)	Accounting Approach (2)	Markups From Materials Share (3)	Markups From DLEU (2020) (4)	Change in Sales Concentration (5)	Accounting Approach (6)	Markups From Materials Share (7)	Markups From DLEU (2020) (8)
		PAN	VELA CONTROLLIN	NG FOR MAIN FFFF	CT OF MARKLIPS AN	ND CONCENTRATIC	NO	
Task displacement	-1.37	-1.31	-1.42	-1.31	-1.40	-1.34	-1.40	-1.37
	(0.18)	(0.20)	(0.20)	(0.18)	(0.21)	(0.22)	(0.22)	(0.21)
Exposure to changes in	1.87	0.26	-0.77	-0.67	1.40	-0.90	-0.34	-0.67
	(1.43)	(1.44)	(0.43)	(1.00)	(1.50)	(1.37)	(0.43)	(1.08)
Share variance explained by:								
	0.57	0.55	0.59	0.55	0.54	0.52	0.54	0.53
markups/concetration	0.04	-0.00	-0.07	0.01	0.03	0.01	-0.03	0.01
R-squared	0.84	0.84	0.84	0.84	0.83	0.83	0.83	0.83
Observations	500	500	500	500	500	500	500	500
		P	ANEL B. NET OUT M	IARKUPS FROM CO	<b>NSTRUCTION OF TA</b>	SK DISPLACEMEN	Т	
Task displacement	-1.74	-1.71	-1.12	-1.32	-2.11	-2.11	-0.90	-1.31
1	(0.22)	(0.24)	(0.15)	(0.16)	(0.34)	(0.33)	(0.14)	(0.20)
Exposure to changes in	0.69	-0.68	-2.09	-2.13	0.01	-1.71	-0.27	-0.49
markups or concentration	(1 50)	(1.40)	(0.53)	(0.75)	(1 53)	(1 35)	(U 44)	(112)
Share variance explained hy:	(00.11)	(01.11)	(2220)	(21.2)	( 2217 )	(	(	()
-task displacement	0.57	0.56	0.54	0.50	0.48	0.48	0.52	0.53
	0.02	0.01	-0.19	0.03	0.00	0.03	-0.02	0.01
R-squared	0.83	0.83	0.85	0.86	0.82	0.82	0.82	0.82
Observations	500	500	500	500	500	500	500	500
<i>Note:</i> This table presents est markups. These groups are defin A we control for groups' snooid	imates of the relationsl ed by gender, education	hip between task dis n, age, race, and nat	splacement and the ch ive/immigrant status.	The dependent varia	s across 500 demogra ble is the change in ho	phic groups controll ourly wages for each	ling for changes in ma t group between 1980	urket structure and and 2016. In Panel
A, we control for groups' special	Ization in industries wi	th changes in mark	et structure leading to	b higher markups. In	column 1. we proxy (	changes in market s	tructure by rising sale	es concentration in

the industry. In column 2-4, we directly control for changes in markups. These are computed as the ratio of revenue to costs in column 2, the inverse of the materials' share in gross output in column 3, and markups estimated using a production function approach as in De Loecker, Eeckhout Jan, and Unger (2020) in column 4. In Panel B, we also subtract the percent increase in markups from the percent decline in the labor share when computing our measure of task displacement (using the accounting markup in columns 1 and 5). Columns 1–4 report results using our measure of task displacement based on observed labor share declines (net of markups in Panel B). Columns 5–8 report results using our measure of task displacement based on automation-driven labor share declines (net of markups in Panel B). In addition to the covariates reported in the table, all specifications control for industry shifters, baseline wage shares in manufacturing, and dummies for education (for no high-school degree, completed high school, some college, college degree and post-graduate degree) and gender. All regressions are weighted by total hours worked by each group in 1980. Standard ŋ valually set 1, we provy errors robust to heteroskedasticity are reported in parentheses. edno 19

# 2002

#### D. ACEMOGLU AND P. RESTREPO

		Der	endent Variable: Chang	e in Hourly wages 1980–2	016	
	Task Displ	acement Based on Labor	Share Declines	Task Displac	ement Based on Automat Labor Share Declines	on-Driven
	(1)	(2)	(3)	(4)	(5)	(9)
			PANEL A. VARIATION	ACROSS U.S. REGIONS		
Task displacement	-1.60	-1.07	-1.31	-1.66	-1.14	-1.41
-	(0.11)	(0.12)	(0.25)	(0.11)	(0.13)	(0.29)
K-squared Observations	0.62 2633	0.81	0.82 2633	0.01 2633	0.81 2633	0.82
		PANEL B. VARIATION	ON ACROSS U.S. REGION	S ABSORBING NATIONAL	TRENDS BY GROUP	
Task displacement	-1.30	-0.26	-0.37	-1.34	-0.37	-0.51
Demonsol	(0.10)	(0.08)	(0.12)	(0.06)	(0.11)	(0.15)
Observations	2633	2633	2633	2633	2633	2633
			PANEL C. VARIATION AC	ROSS COMMUTING ZONES		
Task displacement	-1.23	-0.94	-1.12	-1.37	-1.23	-1.43
	(0.15)	(0.14)	(0.22)	(0.17)	(0.18)	(0.29)
r-squared Observations	00 20.768	00.0	00.0	00.00 20.768	70.0 16.0	70.0 892.02
CO2CI / 4110113	20,100	PANEL D. VARIATION	ACROSS COMMUTING ZC	NES ABSORBING NATION.	AL TRENDS BY GROUP	20,100
Task displacement	-0.77	-0.42	-0.41	-1.02	-0.58	-0.65
	(0.01)	(0.07)	(0.15)	(0.07)	(0.07)	(0.16)
R-squared	0.71	0.78	0.79	0.72	0.78	0.80
Observations	20,768	20,768	20,702	20,/08	20,768	20,768
Covariates. Industry shifters		/.	7.		, ,	/
manufacturing share,		•	•		•	•
education and gender						
tunnuus Evanceura to labor chara						
declines and relative specialization in routine jobs			>			>
<i>Note:</i> This table presents estimates of the relademographic groups defined by gender, education	ationship between	task displacement and the class 9 Census regions. In Pan	hange in real wages across of els C and D, we focus on 54	lemographic groups × regio demographic groups defined	n cells. In Panels A and B, w.	e focus on 300 nd race across
account for all national trends affecting a specific g	group. Columns 1	Thy wages for each cell betwee -3 report results using our me	asure of task displacement h	D and D, we provide estimation pased on observed labor shar	es controning tor group incu e declines. Columns 4–6 repo	t results using
our measure of task displacement based on autom: shifters, baseline wage shares in manufacturing, re	ation-driven labor egional dummies,	· share declines. In addition to and dummies for education	the covariates reported in t (for no high-school degree,	he table and the panel heade completed high school, som	rs, columns 2–3 and 4–5 contr e college, college degree and	ol tor industry post-graduate

Task displacement and changes in real hourly wages, 1980–2016; regional variation.

TABLE VII

degree) and gender. Columns 3 and 6 control for relative specialization in routine jobs and groups' exposure to industry labor share decline. All regressions are weighted by total hours worked by each group-region cell in 1980. Standard errors robust to heteroskedasticity and correlation within demographic group (in Panels A and B) or commuting zone (in Panels C and D) are reported in parentheses.

## TASKS, AUTOMATION, AND RISE IN U.S. WAGE INEQUALITY

2003

In all specifications and with either measure of task displacement, our results are similar to the baseline estimates presented in Table I. For example, the effects of task displacement range between -1.31 and -1.42, while exposure to changes in concentration or markups have little explanatory power for wages.

Panel B goes one step further and uses a measure of task displacement that partials out the component of industry labor share changes driven by markups.<sup>28</sup> This correction does not affect our conclusions, and our point estimates for the effects of task displacement remain sizable and precise. Even with this correction, markup changes do not have a robust effect on wages and explain no more than 3% of the variation in wage changes.

The findings in this table suggest that our task displacement variable is not picking up confounding effects of changes in markups or concentration. These results imply that task displacement—and not so much rising market power—has played a defining role in the surge in U.S. wage inequality over the last four decades.

Our baseline measure of task displacement equates jobs that can be automated with routine jobs, as defined in Acemoglu and Autor (2011). Table A-V in the Supplemental Material Appendix A reports analogous results when we use other measures of jobs that can be automated. The results are similar when we rely on the measure of automatable jobs constructed by Webb (2020) based on the text of new patents. Table A-VI controls for the exposure of workers to occupations in the bottom tercile of the overall wage distribution and the interaction between this exposure and industry automation. This has no effect on our task displacement results and these variables themselves are not significant. This robustness check confirms that task displacement is capturing the effects of automation on worker groups specializing in tasks that can be automated rather than on workers in low-pay occupations.

#### 4.6. Regional Variation

Task and industry composition vary greatly across regions and commuting zones in the United States. To further test the association between task displacement and wages, we now investigate whether regional variation in task displacement also predicts changes in sub-national wage structures.

Table VII provides estimates that exploit regional differences in specialization patterns. The main difference is that now the unit of observation is given by group-region cells, and we exploit differences in specialization across these cells to construct our task displacement measures. In Panel A, we look at 300 demographic groups defined by gender, education, age, and race across nine U.S. regions (giving us a total of 2633 observations excluding empty cells). The results using both measures of task displacement are similar to those in Table I.

In Panel B, we separate regional and national changes by including a full set of demographic group fixed effects that absorb all national trends affecting a demographic group.

Task displacement<sup>direct</sup><sub>g</sub> = 
$$\sum_{i \in \mathcal{I}} \omega_g^i \cdot \frac{\omega_{gi}^R}{\omega_i^R} \cdot (-d \ln s_i^L - d \ln \mu_i)$$

which is the expression we use in Panel B of Table VI.

<sup>&</sup>lt;sup>28</sup>Appendix B-1 provides an extension of our model to an economy with markups. In this more general case, denoting industry *i*'s markup by  $\mu_i$ , our baseline task displacement measure in equation (14) becomes

Table B-6 in the Online Appendix provides additional specifications that allow markups to have a differential impact on workers in routine jobs. Finally, Table B-7 shows the robustness of the patterns reported here to controlling for exposure to industry labor share declines and relative specialization in routine occupations.

We find negative and significant but smaller effects of task displacement (especially in columns 2–3 and 5–6). These estimates imply that task displacement at the regional level matters and has a precisely-estimated negative impact on wages, which is in line with our theory. However, these results also indicate that local differences in task displacement are not as important as national changes for understanding the evolution of the wage structure.<sup>29</sup>

Panels C and D repeat this exercise for 54 demographic groups defined by a coarser grouping of gender, education, age, and race, but now across 722 U.S. commuting zones (for a total of 20,768 observations). The results are similar to those in Panels A and B.

# 4.7. Further Robustness Checks

The Supplemental Material provides a number of additional checks, all of which support our conclusions. First, in Table A-VII, we provide estimates of the effects of task displacement excluding immigrants, as well as separate estimates for men and women. Second, in Table A-VIII, we present stacked-differences models with two periods, 1980–2000 and 2000–2016, which explore the differential patterns of task displacement between these subperiods. Panel A estimates the same specifications as in Table I, but now using stacked differences, while Panel B allows covariates to have different coefficients in the two subperiods. The results in both panels are similar to, but smaller in some specifications than, those in Table I. In Panel C, we report period-by-period estimates of the effects of task displacement on wage changes and confirm that our estimates are comparable across the 1980–2000 and 2000–2016 periods.<sup>30</sup> Finally, in Table A-IX, we present comparable results when we use labor share data from the BLS, exclude extractive industries, winsorize the labor share changes, or focus only on industries with a declining labor share to construct our measure of task displacement.

## 5. GENERAL EQUILIBRIUM EFFECTS AND QUANTITATIVE ANALYSIS

Our reduced-form evidence documented a strong negative relationship between (direct) task displacement and relative wage changes across worker groups. This evidence misses three general equilibrium effects. First, in our regressions, the common impact of productivity on real wages is in the intercept, making our estimates uninformative about wage *level* changes. Second, although our regressions control for *observed* changes in industry composition, they do not separate industry shifts *induced* by automation, missing one component of the total impact of automation. Third and most importantly, our regression estimates focus on the direct effects of automation via task displacement and do not account for the resulting ripple effects, which also impact the wage structure. In this section, we develop a methodology to quantify the effects of technological changes that accounts for these mechanisms, and for brevity, focus on automation technologies.

<sup>&</sup>lt;sup>29</sup>In the presence of migration and trade across regions, regional task displacement should have a smaller impact than national task displacement. For example, in the limit case where tasks can be traded across regions with no transaction or transport costs (or labor is perfectly mobile), one would expect task displacement to reduce the wages of all workers in a given group by the same amount across all regions. In Panel B, national effects are absorbed by the group fixed effects.

<sup>&</sup>lt;sup>30</sup>In Appendix Table B-8, we confirm our findings for the 1980–2007 period, which avoids any persistent effects of the Great Recession.

#### 5.1. General Equilibrium Effects and the Propagation Matrix

We first generalize Proposition 2 to an economy with multiple sectors and with ripple effects (relaxing Assumption 1 which was imposed when we derived equation (13)). For this purpose, let us define *aggregate task shares* as

$$\Gamma_g(\boldsymbol{w},\boldsymbol{\zeta},\Psi) = \sum_{i\in\mathcal{I}} \underbrace{s_i^Y(\boldsymbol{p},c)\cdot(A_i\cdot p_i)^{\lambda-1}}_{=\zeta_i} \cdot \underbrace{\frac{1}{M_i}\int_{\mathcal{T}_{gi}}\psi_g(x)^{\lambda-1}\cdot dx}_{=\Gamma_{gi}},$$

which are given by a weighted sum of industry-specific task shares,  $\Gamma_{gi}$ . Because worker groups now compete for tasks, task shares in each industry are a function of both wages and technology, and also depend on industry shifters,  $\boldsymbol{\zeta} = (\zeta_1, \dots, \zeta_I)$ .

To characterize ripple effects, consider any technological change with a direct effect of  $z_g$  on the real wage of group g. For example, in the case of automation technologies,  $z_g$  corresponds to the direct task displacement of group g. Denote by z the column vector of  $z_g$ 's and differentiate (2) to obtain

$$d\ln \boldsymbol{w} = \boldsymbol{z} + \frac{1}{\lambda} \frac{\partial \ln \Gamma(\boldsymbol{w}, \boldsymbol{\zeta}, \Psi)}{\partial \ln \boldsymbol{w}} \cdot d\ln \boldsymbol{w} \quad \Rightarrow \quad d\ln \boldsymbol{w} = \underbrace{\left(\mathbb{1} - \frac{1}{\lambda} \frac{\partial \ln \Gamma(\boldsymbol{w}, \boldsymbol{\zeta}, \Psi)}{\partial \ln \boldsymbol{w}}\right)^{-1}}_{\Theta} \cdot \boldsymbol{z},$$

where  $\partial \ln \Gamma(\boldsymbol{w}, \boldsymbol{\zeta}, \Psi) / \partial \ln \boldsymbol{w}$  is the  $G \times G$  Jacobian of the function  $\ln \Gamma(\boldsymbol{w}, \boldsymbol{\zeta}, \Psi) = (\ln \Gamma_1(\boldsymbol{w}, \boldsymbol{\zeta}, \Psi), \ln \Gamma_2(\boldsymbol{w}, \boldsymbol{\zeta}, \Psi), \dots, \ln \Gamma_G(\boldsymbol{w}, \boldsymbol{\zeta}, \Psi))$  with respect to the vector of wages  $\boldsymbol{w}$ . This Jacobian summarizes the impact of a change in wages on task allocation (which was equal to zero under Assumption 1). We refer to the  $G \times G$  matrix  $\Theta$  as the *propagation matrix*. Although this matrix is much lower-dimensional than the full set of task-specific productivity functions (the  $\psi_g$ 's), it fully accounts for ripple effects and the general equilibrium implications of technological changes.

In Appendix A-2, we prove that  $\Theta$  is well defined and has several properties. First,  $\theta_{gg'} \ge 0$  captures the extent of competition for tasks between groups g' and g. Second, we show that the row sum of  $\Theta$ , which we denote by  $\varepsilon_g$ , is always between 0 and 1. Third,  $\Theta$ satisfies the following symmetry property:  $\varepsilon_g - \theta_{gg'}/s_{g'}^L = \varepsilon_{g'} - \theta_{g'g}/s_g^L$  for any two groups gand g' (where  $s_g^L$  is the labor share of group g in aggregate output). Finally, the entries of  $\Theta$  also specify whether different workers are q-complements or q-substitutes: an increase in the supply of workers of type g' reduces the real wage of type g if and only if  $\theta_{gg'} > s_{g'}^L \cdot \varepsilon_g$ (Appendix B-1 relates these entries to common measures of elasticities of substitution). In what follows, we denote the gth row of the propagation matrix by  $\Theta_g = (\theta_{g_1}, \ldots, \theta_{g_G})$ .

The next proposition characterizes the general equilibrium effects of automation on wages, industry prices, TFP, and aggregate output (GDP).

**PROPOSITION 4**—GE Effects: The effects of automation on wages, industry prices, and aggregates are given by the solution to the system of equations:

$$d\ln w_g = \Theta_g \cdot \left(\frac{1}{\lambda}d\ln y + \frac{1}{\lambda}d\ln \zeta - \frac{1}{\lambda}d\ln\Gamma^{\text{auto}}\right) \quad \text{for all } g \in \mathcal{G},$$
$$d\ln \zeta_g = \sum_{i \in \mathcal{I}} \omega_{gi} \cdot \left(\frac{\partial\ln s_i^Y(p)}{\partial\ln p} \cdot d\ln p + (\lambda - 1) \cdot d\ln p_i\right) \quad \text{for all } g \in \mathcal{G},$$

$$d\ln p_{i} = \sum_{g \in \mathcal{G}} s_{gi}^{L} \cdot \left( d\ln w_{g} - d\ln \Gamma_{gi}^{\text{auto}} \cdot \pi_{gi} \right) \quad \text{for all } i \in \mathcal{I},$$
  
$$d\ln \text{tfp} = \sum_{i \in \mathcal{I}} s_{i}^{Y}(p) \sum_{g \in \mathcal{G}} s_{gi}^{L} \cdot d\ln \Gamma_{gi}^{\text{auto}} \cdot \pi_{gi},$$
  
$$d\ln y = \frac{1}{1 - s^{K}} \cdot \left( d\ln \text{tfp} + s^{K} \cdot d\ln s^{K} \right),$$
  
$$d\ln s^{K} = -\frac{1}{s^{K}} \sum_{g \in \mathcal{G}} s_{g}^{L} \cdot \left( d\ln w_{g} - d\ln y \right),$$

where  $d \ln \Gamma^{\text{auto}} = (\sum_{i \in \mathcal{I}} \omega_1^i \cdot d \ln \Gamma_{1i}^{\text{auto}}, \dots, \sum_{i \in \mathcal{I}} \omega_G^i \cdot d \ln \Gamma_{Gi}^{\text{auto}}), \ d \ln \zeta = (d \ln \zeta_1, \dots, d \ln \zeta_G), \ and \ d \ln p = (d \ln p_1, \dots, d \ln p_I).$ 

As before, real wage changes depend on the productivity effect  $(d \ln y)$ , the induced shifts in industry composition  $(d \ln \zeta)$ , and the (direct) task displacement experienced by all groups  $(d \ln \Gamma^{auto})$ . In the presence of ripple effects, these direct impacts are premultiplied by the *g*th row of the propagation matrix  $\Theta_g$ . Intuitively, because of ripple effects, wage changes for a group depend on whether other groups that compete for the same tasks are being displaced from their tasks. The *g*th row of the propagation matrix,  $\Theta_g$ , has all the necessary information for computing the effects on group *g* from the task displacement experienced by other groups.<sup>31</sup>

The proposition also shows that we can compute the full general equilibrium impact of automation on wages, industry shares and prices, TFP, GDP, and the capital share by solving the above system of equations. The solution to this system will be a function of the same objects we emphasized in Section 2: direct task displacement experienced by different groups (the  $d \ln \Gamma_{gi}^{auto}$ 's) and cost savings from automation (the  $\pi_{gi}$ 's). In addition, we now need two more ingredients. First, it is necessary to specify the full demand system across industries to determine how technological changes impact industry composition, which will in turn affect the wage structure. Second, we have to parameterize and estimate the propagation matrix to account for the endogenous reallocation of tasks in response to automation and the resulting ripple effects. The formulas in this proposition clarify that the direct task displacement generated by automation is the (exogenous) impulse that leads to a change in task shares, while ripple effects encoded in the propagation matrix and changes in industry composition determine the full general equilibrium implications of this task displacement.<sup>32</sup>

<sup>&</sup>lt;sup>31</sup>Ripple effects do not affect the expressions for TFP, GDP, or industry prices. This is thanks to the envelope theorem: in an efficient economy, induced worker reallocation has only second-order effects on TFP and industry prices, even though it has a first-order impact on labor demand and wages.

<sup>&</sup>lt;sup>32</sup>This way of quantifying the full general equilibrium effects of technological change is a convenient alternative to approaches based on a full parameterization of the model and numerical computation of its equilibria. In our model, a full parameterization would require information on the comparative advantage schedules of 500 different demographic groups across all tasks and industries, which would have been challenging given the available data. Our alternative approach specifies a much lower-dimensional object, the propagation matrix, and enables us to obtain all of the economically relevant quantities determining the general equilibrium effects of automation (and any other direct shock to task shares) without estimating these schedules. In addition, we show that the propagation matrix can be estimated from variation in wages in response to other groups' task displacement. The drawback is that this approach is exact for small changes and is an approximation when there are larger changes.

#### 5.2. Parameterization, Calibration, and Estimation

Measuring Task Displacement and Cost Savings From Automation. In this section, we use our more general measures of task displacement which only require Assumption 2. We also relax the assumption that  $\lambda = 1$ . In this case, automation-driven task displacement experienced by group g in industry i can be measured as

$$d\ln\Gamma_{gi}^{\text{auto}} = \frac{\omega_{gi}^{R}}{\omega_{i}^{R}} \cdot \frac{-d\ln s_{i}^{L,\text{auto}}}{1 + (\lambda - 1) \cdot s_{i}^{L} \cdot \pi_{i}},$$
(16)

and the (direct) task displacement for this group becomes

Task displacement<sup>direct</sup><sub>g</sub> = 
$$\sum_{i \in \mathcal{I}} \omega_g^i \cdot \frac{\omega_{gi}^R}{\omega_i^R} \cdot \frac{-d \ln s_i^{L,\text{auto}}}{1 + (\lambda - 1) \cdot s_i^L \cdot \pi_i}$$
, (17)

where recall that  $d \ln s_i^{L,\text{auto}}$  denotes the percent change in the labor share driven by automation in industry *i*. The term  $1 + (\lambda - 1) \cdot s_i^L \cdot \pi_i$  in the denominator adjusts for the substitution toward automated tasks following a cost reduction of  $\pi_i$ .

We continue to present results using both measures of task displacement, but for brevity, will focus on our preferred strategy that relies on the component of labor share declines driven by automation. Appendix A-3 shows that when  $\lambda \neq 1$  and there are ripple effects, this component can be estimated from an industry-level regression of the *adjusted decline* in the labor share,  $-d \ln s_i^L - s_i^K \cdot (1 - \sigma_i) \cdot (d \ln w_i - d \ln R_i)$ , on our measures of industry automation. Here,  $\sigma_i$  is the elasticity of substitution between capital and labor in industry *i*, and  $d \ln w_i$  and  $d \ln R_i$  are observed factor price changes facing industry *i*. The term  $s_i^K \cdot (1 - \sigma_i) \cdot (d \ln w_i - d \ln R_i)$  corrects for the endogenous changes in labor share due to factor prices and isolates the direct impact of automation on the labor share (see also Grossman and Oberfield (2021)). In the text, we also set  $\sigma_i = 1$ ,  $\lambda = 0.5$ , and  $\pi_{gi} = \pi_i = 30\%$ , and explore robustness to variations in each one of these parameters in Table A-XI).<sup>33</sup>

Industry Demand. We use a simple CES demand system across industries as in footnote 11:  $s_i^Y(\mathbf{p}) = \alpha_i \cdot p_i^{1-\eta}$ . Following Buera, Kaboski, and Rogerson (2015), we set the elasticity of substitution between industries to  $\eta = 0.2$ .

*Propagation Matrix.* Motivated by the symmetry property of the propagation matrix, we parameterize the extent of competition for tasks between two demographic groups g

<sup>&</sup>lt;sup>33</sup>Our baseline measures of task displacement in Section 4 set  $\lambda = 1$ . This is consistent with our choice of the elasticity of substitution between capital and labor,  $\sigma_i = 1$ , here, because under Assumption 1,  $\sigma_i = \lambda$ . Thus, in both cases, we are setting the elasticity of substitution between capital and labor to unity. In the Supplemental Material, we explore the robustness of our results to  $\sigma_i = 0.8$  and  $\sigma_i = 1.2$ , which is consistent with the range of elasticity of substitution estimates in (Karabarbounis and Neiman (2013), Oberfield and Raval (2020)). The estimate  $\lambda = 0.5$  comes from Humlum (2020), and in the Supplemental Material, we show robustness to  $\lambda = 0.3$  and  $\lambda = 0.7$ . Finally, 30% cost savings from automation are in line with the estimates for industrial robots surveyed in Acemoglu and Restrepo (2020). Although cost savings may differ across technologies and industries, we do not have data to estimate such differences. We additionally report estimates from our model with endogenous labor supply in Appendix B-1.

and g' as a function of their distance (dissimilarity) across  $n \in \mathcal{N}$  dimensions. In particular, we assume that

$$\theta_{gg'} = \frac{1}{2} (\varepsilon_g - \varepsilon_{g'}) \cdot s_{g'}^L + \sum_{n \in \mathcal{N}} \beta_n \cdot f(d_{gg'}^n) \cdot s_{g'}^L \quad \text{for all } g' \neq g \text{ and } \theta_{gg} = \theta \text{ for all } g,$$

where f is a decreasing function of the distance along a given dimension n between groups g' and g, denoted here by  $d_{gg'}^n$ . The assumption of common diagonal term is consistent with our reduced-form analysis, which did not find evidence of significant heterogeneities in the effects of task displacement across groups. The parameter  $\beta_n \ge 0$  gives the importance of dimension n in mediating ripple effects. We choose the following dimensions along which we measure distance between groups: occupational and industry employment shares (which account for overlaps in the types of tasks performed) and education by age (which allows for the possibility that, among workers with or without college, workers of similar ages might be more substitutable than those of different ages; see Card and Lemieux, 2001).

Using this parameterization, wage changes from Proposition 4 can be written as

$$d \ln w_{g} = \frac{\varepsilon_{g}}{\lambda} \cdot d \ln y - \frac{\theta}{\lambda} \cdot \text{Task displacement}_{g}^{\text{direct}} - \sum_{g' \neq g} \left( \frac{1}{2} \left( \frac{\varepsilon_{g}}{\lambda} - \frac{\varepsilon_{g'}}{\lambda} \right) + \sum_{n \in \mathcal{N}} \frac{\beta_{n}}{\lambda} \cdot f(d_{g,g'}^{n}) \right) \cdot s_{g'}^{L} \cdot \text{Task displacement}_{g'}^{\text{direct}} + u_{g},$$
  
subject to : $\varepsilon_{g} = \theta + \sum_{g' \neq g} \left( \frac{1}{2} (\varepsilon_{g} - \varepsilon_{g'}) + \sum_{n \in \mathcal{N}} \beta_{n} \cdot f(d_{g,g'}^{n}) \right) \cdot s_{g'}^{L}, \text{ and } \beta_{n} \ge 0,$ 

where the second line represents the ripple effects, f is chosen as an inverted sigmoid function of the distance between two groups, and the error term  $u_g$  is derived from the unobserved wage effects for group g, denoted by  $v_g$  in (15). Specifically, using vector notation,  $\boldsymbol{u} = \boldsymbol{\Theta} \cdot \boldsymbol{v}$ . To estimate the parameters of this system, we impose the exclusion restriction that each group's (automation-driven) task displacement, Task displacement $g_g^{\text{direct}}$ , is orthogonal to  $\boldsymbol{u}$ , or equivalently to  $\boldsymbol{v}$ , which leads to the moment conditions;<sup>34</sup>

$$\mathbb{E}\left[\upsilon_{g} \cdot \left(1, \text{Task displacement}_{g}^{\text{direct}}, \left\{\sum_{g' \neq g} f\left(d_{g,g'}^{n}\right) \cdot s_{g'}^{L} \cdot \text{Task displacement}_{g'}^{\text{direct}}\right\}_{n \in \mathcal{N}}\right)\right] = 0$$
  
for  $g = 1, \dots, G$ .

$$f(d_{g,g'}^n) = \frac{1}{1 + (1/d_{g,g'}^n - 1)^{-\kappa}},$$

2009

<sup>&</sup>lt;sup>34</sup>We compute distances using the dissimilarity measures  $d_{g,g'}^{\text{occupations}} = \frac{1}{2} \sum_{o} |\omega_{g}^{o} - \omega_{g'}^{o}|$  and  $d_{g,g'}^{\text{industries}} = \frac{1}{2} \sum_{i \in \mathcal{I}} |\omega_{g}^{i} - \omega_{g'}^{i}|$ , where the sum runs over 330 occupations and 192 industries in the U.S. Census, respectively. In addition, the sigmoid function takes the form

where  $\kappa \ge 1$  is a tuning parameter governing the decay of the function. For  $\kappa = 1$ , we get f(d) = 1 - d. More generally, the sigmoid function has a maximum of 1 when there is no dissimilarity between two groups. In our baseline estimates in Table A-X, we use a quadratic tuning parameter,  $\kappa = 2$ . In the Supplemental Material, we provide analogous estimates for different values of the tuning parameter; see Tables A-10 and A-XI.

Table A-X in the Supplemental Material provides GMM estimates for  $\theta$  and  $\beta_n$  based on these moment conditions. Columns 1–3 use the task displacement measure based on the observed industry labor share declines, while columns 4–6 use our task displacement measure based on the automation-driven labor share declines. We find positive and significant estimates for ripple effects by occupation, industry, and within age ×education cells. These estimates imply that demographic groups that are directly displaced by automation then compete for tasks performed by other groups that have similar age and education and that specialize in similar occupations and industries.

# 5.3. General Equilibrium Estimates

This subsection presents general equilibrium estimates of the consequences of automation. We use Proposition 4 to compute the full general equilibrium effects of (direct) task displacement, which we measure using equations (16) and (17).

Table VIII summarizes our findings. The first column depicts the data, while the second column presents our general equilibrium estimates when direct task displacement is given by our measure based on observed labor share declines. The third column provides our preferred estimates, feeding in direct task displacement numbers based on the automation-driven component of industry labor share declines. In what follows, we focus on the results in this last column.

The first panel of the table summarizes the effects of automation on the wage structure. This information is also displayed in Figure 7, which decomposes the contribution of the different mechanisms via which task displacement affects wages (with demographic groups sorted by their baseline wage in 1980 on the horizontal axis). Panel A of the figure plots the common productivity effect,  $(1/\lambda) \cdot d \ln y$ , which raises the wages for all groups by close to 45%.

Panel B adds changes in industry composition induced by automation,  $(1/\lambda) \cdot d \ln \zeta$ . Because  $\eta < 1$ , automation induces a shift toward sectors, such as services, that have less automation and increases the demand for workers specialized in these sectors. However, this effect is modest, accounting for less than 7% of the observed changes in the U.S. wage structure.

Panel C adds the direct task displacement effects, given by  $-(1/\lambda) \cdot d \ln \Gamma^{\text{auto}}$ . Direct task displacement generates sizable dispersion in wage changes and causes as much as a 25% decline in the real wages of some groups. The comparison between Panels B and C shows that the main impact of automation on the wage structure is via direct task displacement. This reiterates that automation is distinct from trade in final goods and from other technologies that do not generate task displacement and impact labor demand mainly by changing industry composition.

The results in Panel C show that direct task displacement accounts for as much as 94% of the overall changes in the U.S. wage structure between 1980 and 2016 (see the second row of Table VIII). The reason why this is larger than the 50–70% estimate we obtained in Section 4 is instructive. Our reduced-form analysis did not allow for ripple effects, which were thus partially captured by our task displacement measures. Our general equilibrium framework clarifies that ripple effects enable directly-displaced demographic groups to compete for non-automated tasks performed by other groups and hence spread the impacts of task displacement across groups. This is confirmed in Panel D of Table VIII, which depicts the full effects of automation on wages after accounting for ripple effects using our estimates of the propagation matrix. For example, the direct impact of automation on high-school graduate white men aged 26–35 in Panel C of Figure 7 is -13.3%.

#### TABLE VIII General equilibrium effects.

	Data for 1980–2016 (1)	Using Task Displacement Based on Labor Share Declines (2)	Using Task Displacement Based on Automation- Driven Labor Share Declines (3)
WAGE STRUCTURE:			
Share wage changes explained:			
-due to changes in industry composition		6.78%	6.33%
-adding direct displacement effects		100.54%	93.34%
-accounting for ripple effects		48.35%	46.88%
Rise in college premium	25.51%	21.82%	21.02%
-part due to direct displacement effect		40.92%	37.71%
Rise in post-graduate premium	40.42%	24.06%	22.42%
-part due to direct displacement effect		48.04%	43.57%
Change in gender gap	15.37%	1.83%	1.90%
<ul> <li>—part due to direct displacement effect</li> </ul>		6.31%	5.94%
Share with declining wages	53.10%	41.71%	42.26%
<ul> <li>—part due to direct displacement effects</li> </ul>		49.61%	51.52%
Wages for men with no high school	-8.21%	-7.18%	-8.41%
<ul> <li>—part due to direct displacement effects</li> </ul>		-13.97%	-15.11%
Wages for women with no high school	10.94%	1.24%	-3.40%
-part due to direct displacement effects		6.21%	-2.82%
AGGREGATES:			
Change in average wages, $d \ln w$	29.15%	5.71%	5.18%
Change in GDP per capita, $d \ln y$	70.00%	23.42%	20.95%
Change in TFP, $d \ln tfp$	35%	3.77%	3.42%
Change in labor share, $ds^L$	-8 p.p	-11.69 p.p	-10.41 p.p
Change in $K/Y$ ratio	30.00%	41.93%	38.10%
SECTORAL PATTERNS:			
Share manufacturing in GDP	-8.80 p.p	-0.41 p.p	-0.52 p.p
Change in manufacturing wage bill	-35.00%	$-8.23\overline{\%}$	-12.85%
(per capita)			

*Note*: This table summarizes the effects of task displacement on the wage distribution, real wage levels, aggregates, and industry outcomes. These are computed using the formulas in Proposition 4 and the parameterization and estimates for the industry demand system and the propagation matrix in Section 5.2. Column 2 computes the model predictions based on our measure of task displacement from industry labor share declines, while column 3 computes the model predictions based on our measure of task displacement from automation-driven labor share declines. The wage data reported in column 1 are from the 1980 U.S. Census and 2014–2018 ACS. The data for GDP, the labor share, the capital-output ratio data, and the industry patterns for manufacturing are from the BEA and the BLS. The TFP data are from Fernald (2014).

But once we allow for ripple effects in Panel D of Figure 7, this demographic group experiences a smaller, 5.5%, decline in real wages. In contrast, the direct impact on Hispanic high-school dropout women aged 36–45 is a 2.6% real wage increase, but incorporating ripple effects, this group suffers a 3.3% real wage decline.

As a summary, Figure 8 plots the predicted wage changes in the model and the observed real wage changes between 1980 and 2016. In addition to accounting for a large fraction of the variation in U.S. wage structure, automation explains several other salient aspects of the labor market during this period. First, even though there is a large (close to 45%) productivity effect, real wages for 131 demographic groups (making up 42% of the 1980 population) *decline* because of automation (in the data, 121 groups, making up 53% of the



FIGURE 7.—Contribution of productivity effects, industry shifts, direct displacement effects, and ripple effects to the predicted change in hourly wages, 1980–2016. Marker sizes indicate the share of hours worked by each group and shades of gray (different colors) indicate education levels. See text for variable definitions.

1980 population, experienced real wage declines). This result highlights how automation can generate meaningful real wage declines, which contrasts with the canonical SBTC model, where technological *progress* is predicted to increase the real wages of all groups. Second, in general equilibrium, task displacement generates a 21% increase in the college premium (80% of the observed increase) and a 22% increase in the post-graduate premium (55% of the observed increase). Finally, task displacement alone closes the gender gap by about 2%. Interestingly, in all these cases, the direct effects of task displacement are dampened once we account for ripple effects. For example, the direct effect of automation is to reduce the gender gap by 6%, but because displaced men compete for tasks previously performed by women, in general equilibrium the gender premium declines only by 2%.

Despite matching several salient aspects of the changes in the U.S. wage structure, our model misses a significant portion of wage growth for highly-educated workers at the top of the wage distribution. This may reflect the complementarity between some of the new technologies and post-graduate workers or other forces, such as winner-take-all dynamics in some high-skill professions, which are both absent from our model.

The second panel of Table VIII turns to the model's implications for aggregates. Despite the large distributional effects documented above, task displacement generates only a cumulative 3.4% TFP gain over 1980–2016, and this is the reason why average real wages are predicted to grow slowly (only by 5.2%) and many groups experience real wage declines. This small TFP increase is intuitive in light of the characterization in Proposition 4: TFP gains from automation can be approximated as the product of the share in GDP of displaced tasks ( $\sum_g \sum_i s_{gi}^L \cdot d \ln \Gamma_{gi}^{auto}$ ), which is approximately 10%, and average cost reductions of 30%, thus yielding a  $0.1 \times 0.3 \approx 3\%$  increase in TFP. In contrast to this small automation-induced increase in productivity, in the data TFP grew by 35% during this period, and average real wages rose by 29% (though two thirds of the latter is due to educational upgrading of the workforce, which is not present in our model). These numbers confirm that there were other technological advances—such as factor-augmenting and productivity-deepening technologies, industry TFP, or even new tasks—contributing to GDP, wage growth, and productivity between 1980 and 2016. However, the congruence between the model-implied changes in wage structure and the data suggests that



FIGURE 8.—Observed wage changes (vertical axis) versus predicted wage changes in general equilibrium due to automation (horizontal axis). the 45° line is shown in black. Marker sizes indicate the share of hours worked by each group and shades of gray (different colors) indicate education levels. See text for variable definitions.

these other technological changes had small effects on inequality, except possibly at the top of the wage distribution. Finally, task displacement due to automation accounts for the observed decline in the labor share (by construction) and the observed increase in the capital-GDP ratio over this period. This last finding implies that the amount of investment accompanying automation in our model is in the ballpark of the data.

The third panel of Table VIII summarizes the industry implications of task displacement. In line with the modest TFP gains estimated above, we see that task displacement generates small changes in industry composition and accounts for only 0.5 of the 8.8 percentage point decline in the share of manufacturing in GDP. Despite its small impact on industry composition, task displacement within manufacturing generates a large, 13%, reduction in the wage bill of that sector, accounting for a third of the decline in manufacturing labor demand for 1980–2016.

## 6. CONCLUDING REMARKS

This paper argued that a significant portion of the rise in U.S. wage inequality over the last four decades has been driven by automation (and to a lesser extent offshoring), which displaces certain worker groups from employment opportunities for which they had comparative advantage. To develop this point, we proposed a conceptual framework where tasks are allocated to different types of labor and capital, and automation technologies expand the set of tasks performed by capital at the expense of workers previously employed in these tasks. We derived a simple equation linking wage changes of a demographic group to the task displacement it experiences.

Our reduced-form evidence is based on estimating this equation and reveals a number of striking new facts. Most notably, we documented that 50-70% of the changes in the U.S. wage structure between 1980 and 2016 are accounted for by the relative wage

2013

declines of worker groups specialized in routine tasks in industries experiencing rapid automation. We also verified that our task displacement variable captures the effects of automation technologies (and to a lesser degree offshoring) rather than changes in overall capital intensity, other types of technologies, markups, industry concentration, unionization, or Chinese import competition. These alternative economic trends do not appear to play a major role in the evolution of the U.S. wage structure between 1980 and 2016 and have negligible effects on our estimates.

Our reduced-form regressions focus on the direct effects of task displacement on wages, but miss important general equilibrium forces. We developed a methodology to quantify the general equilibrium implications of task displacement, which can account for the implications of automation working through productivity gains, ripple effects and changes in industry composition. Our full quantitative evaluation shows that task displacement explains close to 50% of the observed changes in the U.S. wage structure. Most notably, task displacement leads to sizable increases in wage inequality, but only small productivity gains—thus providing a possible resolution to a puzzling feature of the U.S. data.

There are several interesting areas for future research. First, our framework has been static, and any effects from capital accumulation, dynamic incentives for the development of new technologies, and education and skill acquisition are absent. Incorporating these effects is an important direction for future research.

Second, we did not attempt to model and estimate the effects of technologies introducing new labor-intensive tasks (which we argued to have been important in previous work, Acemoglu and Restrepo (2018)). This is another avenue for future research.

Third, our strategy exploited industry-level trends in automation and labor share. Several recent works have pointed out that labor share declines concentrate on a subset of, often largest, firms (e.g., Autor et al. (2020), Kehrig and Vincent (2020)). Acemoglu, Lelarge, and Restrepo (2020) showed that in French manufacturing, these are the firms that adopt automation technologies and expand at the expense of their competitors, where the actual declines in labor demand take place. This pattern confirms that it is (automation-driven) reductions in the labor share at the industry level, rather than at the firm level, that are relevant for task displacement, but also suggests that modeling the competition between automating and non-automating firms is yet another interesting area for future research (see, e.g., Hubmer and Restrepo (2021)).

Finally, our empirical work has been confined to the United States and the 1980–2016 period, for which we have all the data components necessary for our reduced-form and quantitative analyses. Expanding these data sources and the empirical exploration of the role of task displacement to earlier periods and other economies is an important direction for research that may help us understand the technological and institutional reasons why the U.S. wage structure was quite stable for the three decades leading up to the mid-1970s.

#### REFERENCES

ACEMOGLU, DARON, AND PASCUAL RESTREPO (2018): "The Race Between Man and Machine: Implications of Technology for Growth, Factor Shares, and Employment," *American Economic Review*, 108, 1488–1542. [1976,2014]

(2020): "Robots and Jobs: Evidence From US Labor Markets," *Journal of Political Economy*, 128, 2188–2244. [1975,1976,1985-1987,2008]

(2022): "Supplement to 'Tasks, Automation, and the Rise in U.S. Wage Inequality'," *Econometrica Supplemental Material*, 90, https://doi.org/10.3982/ECTA19815. [1977]

ACEMOGLU, DARON, AND DAVID AUTOR (2011): "Skills, Tasks and Technologies: Implications for Employment and Earnings," in *Handbook of Labor Economics*, Vol. 4, 1043–1171. [1973,1975,1976,1983,1989, 2004]

- ACEMOGLU, DARON, AND FABRIZIO ZILIBOTTI (2001): "Productivity Differences\*," *The Quarterly Journal of Economics*, 116, 563–606. [1976]
- ACEMOGLU, DARON, DAVID AUTOR, JONATHON HAZELL, AND PASCUAL RESTREPO (2022): "Artificial Intelligence and Jobs: Evidence from Online Vacancies," *Journal of Labor Economics*, 40, S293–S340. [1986]
- ACEMOGLU, DARON, CLAIRE LELARGE, AND PASCUAL RESTREPO (2020): "Competing With Robots: Firm-Level Evidence From France," in *AEA Papers and Proceedings*, Vol. 110, 383–388. [1975,2014]
- ADAO, RODRIGO, MICHAL KOLESÁR, AND EDUARDO MORALES (2019): "Shift-Share Designs: Theory and Inference," *The Quarterly Journal of Economics*, 134, 1949–2010. [1994]
- ATALAY, ENGHIN, PHAI PHONGTHIENGTHAM, SEBASTIAN SOTELO, AND DANIEL TANNENBAUM (2020): "The Evolution of Work in the United States," *American Economic Journal: Applied Economics*, 12, 1–34. [1977]
- AUTOR, DAVID H. (2019): "Work of the Past, Work of the Future," AEA Papers and Proceedings, 109, 1–32. [1973,1974]
- AUTOR, DAVID H., AND DAVID DORN (2013): "The Growth of Low-Skill Service Jobs and the Polarization of the US Labor Market," *American Economic Review*, 103, 1553–1597. [1976]
- AUTOR, DAVID H., DAVID DORN, AND GORDON H. HANSON (2013): "The China Syndrome: Local Labor Market Effects of Import Competition in the United States," *American Economic Review*, 103, 2121–2168. [1987]
- AUTOR, DAVID H., DAVID DORN, LAWRENCE F. KATZ, CHRISTINA PATTERSON, AND JOHN VAN REENEN (2020): "The Fall of the Labor Share and the Rise of Superstar Firms," *The Quarterly Journal of Economics*, 135, 645–709. [1975,2014]
- AUTOR, DAVID H., LAWRENCE F. KATZ, AND ALAN B. KRUEGER (1998): "Computing Inequality: Have Computers Changed the Labor Market?" *The Quarterly Journal of Economics*, 113, 1169–1213. [1977]
- AUTOR, DAVID H., FRANK LEVY, AND RICHARD J. MURNANE (2003): "The Skill Content of Recent Technological Change: An Empirical Exploration," *The Quarterly Journal of Economics*, 118, 1279–1333. [1976, 1985]
- BÁRÁNY, ZSÓFIA L., AND CHRISTIAN SIEGEL (2020): "Biased Technological Change and Employment Reallocation," *Labour Economics*, 67, 101930. [1977]
- BORUSYAK, KIRILL, PETER HULL, AND XAVIER JARAVEL (2022): "Quasi-Experimental Shift-Share Research Designs," *The Review of Economic Studies*, 89, 181–213. [1994]
- BOUND, JOHN, AND GEORGE JOHNSON (1992): "Changes in the Structure of Wages in the 1980's: An Evaluation of Alternative Explanations," *American Economic Review*, 82, 371–392. [1977]
- BUERA, FRANCISCO J., JOSEPH P. KABOSKI, AND RICHARD ROGERSON (2015): "Skill Biased Structural Change," Tech. rep., National Bureau of Economic Research. [2008]
- BURSTEIN, ARIEL, EDUARDO MORALES, AND JONATHAN VOGEL (2019): "Changes in Between-Group Inequality: Computers, Occupations, and International Trade," *American Economic Journal: Macroeconomics*, 11, 348–400. [1977]
- CARD, DAVID, AND THOMAS LEMIEUX (2001): "Can Falling Supply Explain the Rising Return to College for Younger Men? A Cohort-Based Analysis\*," *The Quarterly Journal of Economics*, 116, 705–746. [1977,1996, 1998,2009]
- CAUNEDO, JULIETA, DAVID JAUME, AND ELISA KELLER (2021): "Occupational Exposure to Capital-Embodied Technical Change," CEPR Discussion Papers 15759, C.E.P.R. Discussion Papers. [1977]
- DAO, MAI CHI, MITALI DAS, AND ZSOKA KOCZAN (2019): "Why Is Labour Receiving a Smaller Share of Global Income?" *Economic Policy*, 34, 723–759. [1975]
- DE LOECKER, JAN, JAN EECKHOUT, AND GABRIEL UNGER (2020): "The Rise of Market Power and the Macroeconomic Implications," *The Quarterly Journal of Economics*, 135, 561–644. [1975,2001,2002]
- ELSBY, MICHAEL W. L., BART HOBIJN, AND AYŞEGÜL ŞAHIN (2013): "The Decline of the US Labor Share," Brookings Papers on Economic Activity, 2013, 1–63. [1975]
- FEENSTRA, ROBERT C., AND GORDON H. HANSON (1999): "The Impact of Outsourcing and High-Technology Capital on Wages: Estimates for the United States, 1979–1990," *The Quarterly Journal of Economics*, 114, 907–940. [1987]
- FERNALD, JOHN (2014): "A Quarterly, Utilization-Adjusted Series on Total Factor Productivity," Tech. rep, Federal Reserve Bank of San Francisco. [2011]
- GOLDIN, CLAUDIA, AND LAWRENCE F. KATZ (2008): *The Race Between Education and Technology*. Cambridge: Harvard University Press. [1973]
- GOOS, MAARTEN, AND ALAN MANNING (2007): "Lousy and Lovely Jobs: The Rising Polarization of Work in Britain," *The Review of Economics and Statistics*, 89, 118–133. [1976]
- GOOS, MAARTEN, ALAN MANNING, AND ANNA SALOMONS (2014): "Explaining Job Polarization: Routine-Biased Technological Change and Offshoring," *American Economic Review*, 104, 2509–2526. [1976]

- GRAETZ, GEORG, AND GUY MICHAELS (2018): "Robots at Work," *The Review of Economics and Statistics*, 100, 753–768. [1976]
- GREGORY, TERRY, ANNA SALOMONS, AND ULRICH ZIERAHN (2018): "Racing With or Against the Machine? Evidence From Europe," Working Paper No. 7247, CESifo Working Paper. [1977]
- GRILICHES, ZVI (1969): "Capital-Skill Complementarity," *The Review of Economics and Statistics*, 51, 465–468. [1977]
- GROSSMAN, GENE M., AND EZRA OBERFIELD (2021): "The Elusive Explanation for the Declining Labor Share," Working Paper 29165, National Bureau of Economic Research. [2008]
- GROSSMAN, GENE M., AND ESTEBAN ROSSI-HANSBERG (2008): "Trading Tasks: A Simple Theory of Offshoring," American Economic Review, 98, 1978–1997. [1976,1980]
- HUBMER, JOACHIM (2020): "The Race Between Preferences and Technology," Unpublished manuscript, University of Pennsylvania. [1975]
- HUBMER, JOACHIM, AND PASCUAL RESTREPO (2021): "Not a Typical Firm: The Joint Dynamics of Firms, Labor Shares, and Capital-Labor Substitution," Working Paper 28579, National Bureau of Economic Research. [1975,2014]
- HUMLUM, ANDERS (2020): "Robot Adoption and Labor Market Dynamics," Working paper, University of Chicago. [1986,2008]
- JACKSON, MATTHEW O., AND ZAFER KANIK (2020): "How Automation that Substitutes for Labor Affects Production Networks, Growth, and Income Inequality," Unpublished manuscript, Glascow University. [1976]
- JAIMOVICH, NIR, ITAY SAPORTA-EKSTEN, HENRY E. SIU, AND YANIV YEDID-LEVI (2020): "The Macroeconomics of Automation: Data, Theory, and Policy Analysis," Tech. rep., National Bureau of Economic Research. [1977]
- JOHNSON, MATTHEW, AND MICHAEL P. KEANE (2013): "A Dynamic Equilibrium Model of the US Wage Structure, 1968–1996," Journal of Labor Economics, 31, 1–49. [1977]
- KARABARBOUNIS, LOUKAS, AND BRENT NEIMAN (2013): "The Global Decline of the Labor Share," *The Quarterly Journal of Economics*, 129, 61–103. [1975,2008]
- KATZ, LAWRENCE F, AND KEVIN M. MURPHY (1992): "Changes in Relative Wages, 1963–1987: Supply and Demand Factors," The Quarterly Journal of Economics, 107, 35–78. [1977,1983,1996]
- KEHRIG, MATTHIAS, AND NICOLAS VINCENT (2020): "The Micro-Level Anatomy of the Labor Share Decline," NBER Working Papers 25275, National Bureau of Economic Research, Inc. [2014]
- KIM, RYAN, AND JONATHAN VOGEL (2021): "Trade Shocks and Labor Market Adjustment," American Economic Review: Insights, 3, 115–130. [1998]
- KLENOW, PETER J., AND ANDRÉS RODRÍGUEZ-CLARE (1997): "The Neoclassical Revival in Growth Economics: Has It Gone Too Far?" *NBER Macroeconomics Annual*, 12, 73–103. [1993]
- KRUEGER, ALAN B. (1993): "How Computers Have Changed the Wage Structure: Evidence From Microdata, 1984-1989," The Quarterly Journal of Economics, 108, 33–60. [1977]
- KRUSELL, PER, LEE E. OHANIAN, JOSÉ-VÍCTOR RÍOS-RULL, AND GIOVANNI L. VIOLANTE (2000): "Capital-Skill Complementarity and Inequality: A Macroeconomic Analysis," *Econometrica*, 68, 1029–1053. [1977]
- LEE, SANG YOON TIM, AND YONGSEOK SHIN (2017): "Horizontal and Vertical Polarization: Task-Specific Technological Change in a Multi-Sector Economy," Working Paper 23283, National Bureau of Economic Research. [1977]
- NEWEY, WHITNEY K. (1984): "A Method of Moments Interpretation of Sequential Estimators," *Economics Letters*, 14, 201–206. [1994]
- OBERFIELD, EZRA, AND DEVESH RAVAL (2020): "Micro Data and Macro Technology," Econometrica. [2008]
- PIKETTY, THOMAS (2014): Capital in the Twenty-First Century. Cambridge: Harvard University Press. [1975]
- WEBB, MICHAEL (2020): "The Impact of Artificial Intelligence on the Labor Market," Working paper, Stanford University. [2004]
- ZEIRA, JOSEPH (1998): "Workers, Machines, and Economic Growth," *The Quarterly Journal of Economics*, 113, 1091–1117. [1976]

Co-editor Charles I. Jones handled this manuscript.

Manuscript received 7 June, 2021; final version accepted 18 February, 2022; available online 12 March, 2022.